IAEA TECDOC SERIES

IAEA-TECDOC-2019

Establishing a Nuclear Forensic Capability: Application of Analytical Techniques



IAEA SAFETY STANDARDS AND RELATED PUBLICATIONS

IAEA SAFETY STANDARDS

Under the terms of Article III of its Statute, the IAEA is authorized to establish or adopt standards of safety for protection of health and minimization of danger to life and property, and to provide for the application of these standards.

The publications by means of which the IAEA establishes standards are issued in the IAEA Safety Standards Series. This series covers nuclear safety, radiation safety, transport safety and waste safety. The publication categories in the series are Safety Fundamentals, Safety Requirements and Safety Guides.

Information on the IAEA's safety standards programme is available at the IAEA Internet site

www.iaea.org/resources/safety-standards

The site provides the texts in English of published and draft safety standards. The texts of safety standards issued in Arabic, Chinese, French, Russian and Spanish, the IAEA Safety Glossary and a status report for safety standards under development are also available. For further information, please contact the IAEA at: Vienna International Centre, PO Box 100, 1400 Vienna, Austria.

All users of IAEA safety standards are invited to inform the IAEA of experience in their use (e.g. as a basis for national regulations, for safety reviews and for training courses) for the purpose of ensuring that they continue to meet users' needs. Information may be provided via the IAEA Internet site or by post, as above, or by email to Official.Mail@iaea.org.

RELATED PUBLICATIONS

The IAEA provides for the application of the standards and, under the terms of Articles III and VIII.C of its Statute, makes available and fosters the exchange of information relating to peaceful nuclear activities and serves as an intermediary among its Member States for this purpose.

Reports on safety in nuclear activities are issued as **Safety Reports**, which provide practical examples and detailed methods that can be used in support of the safety standards.

Other safety related IAEA publications are issued as **Emergency Preparedness and Response** publications, **Radiological Assessment Reports**, the International Nuclear Safety Group's **INSAG Reports**, **Technical Reports** and **TECDOCs**. The IAEA also issues reports on radiological accidents, training manuals and practical manuals, and other special safety related publications.

Security related publications are issued in the IAEA Nuclear Security Series.

The IAEA Nuclear Energy Series comprises informational publications to encourage and assist research on, and the development and practical application of, nuclear energy for peaceful purposes. It includes reports and guides on the status of and advances in technology, and on experience, good practices and practical examples in the areas of nuclear power, the nuclear fuel cycle, radioactive waste management and decommissioning.

ESTABLISHING A NUCLEAR FORENSIC CAPABILITY: APPLICATION OF ANALYTICAL TECHNIQUES

The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ANTIGUA AND BARBUDA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAIJAN BAHAMAS BAHRAIN BANGLADESH BARBADOS BELARUS BELGIUM BELIZE BENIN BOLIVIA, PLURINATIONAL STATE OF BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BRUNEI DARUSSALAM **BULGARIA BURKINA FASO** BURUNDI CAMBODIA CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA COMOROS CONGO COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DJIBOUTI DOMINICA DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA **ESTONIA** ESWATINI **ETHIOPIA** FLII FINLAND FRANCE GABON GEORGIA

GERMANY GHANA GREECE GRENADA **GUATEMALA** GUYANA HAITI HOLY SEE HONDURAS HUNGARY ICELAND INDIA INDONESIA IRAN, ISLAMIC REPUBLIC OF IRAQ IRELAND ISRAEL ITALY JAMAICA JAPAN JORDAN KAZAKHSTAN **KENYA** KOREA, REPUBLIC OF **KUWAIT** KYRGYZSTAN LAO PEOPLE'S DEMOCRATIC REPUBLIC LATVIA LEBANON LESOTHO LIBERIA LIBYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIQUE MYANMAR NAMIBIA NEPAL NETHERLANDS NEW ZEALAND NICARAGUA NIGER NIGERIA NORTH MACEDONIA NORWAY OMAN PAKISTAN

PALAU PANAMA PAPUA NEW GUINEA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL QATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION RWANDA SAINT KITTS AND NEVIS SAINT LUCIA SAINT VINCENT AND THE GRENADINES SAMOA SAN MARINO SAUDI ARABIA SENEGAL SERBIA SEYCHELLES SIERRA LEONE SINGAPORE SLOVAKIA **SLOVENIA** SOUTH AFRICA SPAIN SRI LANKA SUDAN **SWEDEN** SWITZERLAND SYRIAN ARAB REPUBLIC TAJIKISTAN THAILAND TOGO TONGA TRINIDAD AND TOBAGO TUNISIA TÜRKİYE TURKMENISTAN UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY UZBEKISTAN VANUATU VENEZUELA, BOLIVARIAN REPUBLIC OF VIET NAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

IAEA-TECDOC-2019

ESTABLISHING A NUCLEAR FORENSIC CAPABILITY: APPLICATION OF ANALYTICAL TECHNIQUES

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2023

COPYRIGHT NOTICE

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Marketing and Sales Unit, Publishing Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria fax: +43 1 26007 22529 tel.: +43 1 2600 22417 email: sales.publications@iaea.org www.iaea.org/publications

For further information on this publication, please contact:

Nuclear Security of Materials Outside of Regulatory Control Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria Email: Official.Mail@iaea.org

> © IAEA, 2023 Printed by the IAEA in Austria March 2023

IAEA Library Cataloguing in Publication Data

Names: International Atomic Energy Agency.

Title: Establishing a nuclear forensic capability : application of analytical techniques / International Atomic Energy Agency.

Description: Vienna : International Atomic Energy Agency, 2023. | Series: IAEA TECDOC series, ISSN 1011–4289 ; no. 2019 | Includes bibliographical references.

Identifiers: IAEAL 23-01586 | ISBN 978-92-0-118423-8 (paperback : alk. paper) | ISBN 978-92-0-118323-1 (pdf)

Subjects: LCSH: Nuclear forensics. | Nuclear forensics — Methodology. | Nuclear physics.

FOREWORD

Nuclear forensics supports national nuclear security measures and criminal investigations by providing information on the identity, origin and history of nuclear or other radioactive material found outside of regulatory control. A nuclear forensic examination uses analytical techniques to generate data in the context of a criminal investigation. A nuclear forensic capability can enhance national nuclear security programmes and can support the enforcement of a State's laws that prohibit the possession and use of nuclear or other radioactive material out of regulatory control.

Since 2005, Member States have acknowledged and supported the further development of nuclear forensics at the IAEA in General Conference resolutions. The IAEA programme of assistance in nuclear forensics has grown significantly over the past decade and includes training, coordinated research, technical advisory activities and published guidance. IAEA Nuclear Security Series No. 2-G (Rev. 1), Nuclear Forensics in Support of Investigations, provides guidance to States on implementing a nuclear forensic capability, conducting a nuclear forensic examination and the role of nuclear forensics in criminal investigations and national nuclear security programmes.

This publication takes guidance provided in IAEA Nuclear Security Series No. 2-G (Rev. 1) and provides technical information, including details of analytical techniques and methodologies used in nuclear forensic examinations to support investigations of criminal or other unauthorized acts involving nuclear or other radioactive material.

The preparation of this publication would not have been possible without the contributions of Member States. The IAEA is grateful to all who contributed to the drafting and review of this publication. The IAEA officers responsible for this publication were G. Eppich, D. Smith and E. Széles of the Division of Nuclear Security.

EDITORIAL NOTE

This publication has been prepared from the original material as submitted by the contributors and has not been edited by the editorial staff of the IAEA. The views expressed remain the responsibility of the contributors and do not necessarily represent the views of the IAEA or its Member States.

Neither the IAEA nor its Member States assume any responsibility for consequences which may arise from the use of this publication. This publication does not address questions of responsibility, legal or otherwise, for acts or omissions on the part of any person.

The use of particular designations of countries or territories does not imply any judgement by the publisher, the IAEA, as to the legal status of such countries or territories, of their authorities and institutions or of the delimitation of their boundaries.

The mention of names of specific companies or products (whether or not indicated as registered) does not imply any intention to infringe proprietary rights, nor should it be construed as an endorsement or recommendation on the part of the IAEA.

The authors are responsible for having obtained the necessary permission for the IAEA to reproduce, translate or use material from sources already protected by copyrights.

The IAEA has no responsibility for the persistence or accuracy of URLs for external or third party Internet web sites referred to in this publication and does not guarantee that any content on such web sites is, or will remain, accurate or appropriate.

CONTENTS

1.	INTRODUCTION1		
	1.1.	BACKGROUND1	
	1.2.	OBJECTIVE1	
	1.3.	SCOPE	
	1.4.	STRUCTURE2	
2.	NUCL	LEAR FORENSICS IN SUPPORT OF INVESTIGATIONS OF	
	CRIM	INAL OR OTHER UNAUTHORIZED ACTS INVOLVING NUCLEAR	
	OR O	THER RADIOACTIVE MATERIAL2	
3.	ACTI	VITIES AT A RADIOLOGICAL CRIME SCENE INCLUDING	
	CATE	GORIZATION AND INITIAL CHARACTERIZATION OF NUCLEAR	
	OR O	THER RADIOACTIVE MATERIAL	
Λ	RECE	IPT AND HANDI ING OF SAMPLES AT THE NUCLEAR FORENSIC	
4.	LABC	ORATORY	
	41	RECEIPT OF REQUEST FOR NUCLEAR FORENSIC ANALYSIS	
		FROM THE INVESTIGATIVE AUTHORITY ALONGSIDE	
		RELEVANT IN-FIELD CATEGORIZATION AND	
		CHARACTERIZATION DATA7	
	4.2.	DEVELOPMENT OF A NUCLEAR FORENSIC ANALYTICAL	
		PLAN FOR THE SPECIFIC TYPE OF NUCLEAR OR OTHER	
		RADIOACTIVE MATERIAL	
	4.3.	RECEIPT OF MATERIAL WHILE MAINTAINING CHAIN OF	
	4.4	CUSIODY	
	4.4.	ANALYSIS OF MATERIAL UPON RECEIPT TO CONFIRM ACCUBACY OF IN FIELD CATEGODIZATION	
	15	LINDACKING AND DOCUMENTATION OF MATERIAL AND	
	т.у.	PRESERVATION AND DOCUMENTATION OF MATERIAL, AND	
		ASSOCIATED TRADITIONAL FORENSIC EVIDENCE	
	4.6.	ALIQUOTING FOR SUBSEQUENT NUCLEAR FORENSIC	
		EXAMINATION, IF NECESSARY11	
5.	CHAR	RACTERIZATION THROUGH NON-DESTRUCTIVE ANALYSIS11	
	5 1	PADIOMETRIC NON DESTRUCTIVE ANALYTICAL	
	5.1.	TECHNIQUES 12	
	52	PHYSICAL NON-DESTRUCTIVE MEASUREMENTS 13	
	5.3.	ELEMENTAL AND CHEMICAL NON-DESTRUCTIVE	
		MEASUREMENTS15	
6	CHARACTERIZATION THROUGH DESTRUCTIVE ANALYSIS		
0.	6 1		
	0.1.	ELEIVIENTAL AND CHEIVIICAL COMPOSITION MEASTIDEMENTS 17	
	62	ISOTOPIC COMPOSITION MEASUREMENTS 10	
	6.3	RADIOCHRONOMETRY MEASUREMENTS 21	
	····		

	6.4.	OTHER DESTRUCTIVE ANALYSIS TECHNIQUES	2		
7.	QUALI	TY ASSURANCE IN NUCLEAR FORENSIC ANALYSIS2	23		
8.	REPOR	TING THE RESULTS OF A NUCLEAR FORENSIC EXAMINATION	24		
9.	INTER	NATIONAL COOPERATION AND ASSISTANCE2	26		
REFERENCES					
BIBI	JOGRA	РНҮ4	3		
ANN	EX I: SA FOREN	AMPLES AND GENERAL CONSIDERATIONS FOR NUCLEAR ISIC EXAMINATIONS4	5		
ANN	EX II: D FOREN	DESCRIPTIONS OF ANALYTICAL METHODS USED IN NUCLEAR	51		
ABB	REVIAT	TIONS	59		

1. INTRODUCTION

1.1. BACKGROUND

Nuclear forensic science (hereafter referred to as 'nuclear forensics') is a component of nuclear security and, in particular, of investigations of criminal or other unauthorized acts involving nuclear or other radioactive material. Along with other methods of investigation, nuclear forensics can provide information that may assist investigators in enforcing laws and regulations that prohibit the possession and use of nuclear or other radioactive material out of regulatory control. Nuclear forensics can help identify linkages between nuclear or other radioactive material out of regulatory control and people, places and events subject to the investigation, as well as between disparate criminal activities involving nuclear or other radioactive material. Nuclear forensics may also be used to assist in determining when, where and how the nuclear or other radioactive material was lost from regulatory control. It can aid investigators in the identification of deficiencies and weaknesses in nuclear material accountancy and control systems, as well as vulnerabilities in physical protection systems at nuclear facilities and radiation detection systems at borders, ports and other national points of entry.

A State may perform a nuclear forensic examination, as a component of a broader investigation led by an investigative authority of the State, of the circumstances and criminal aspects of nuclear or other radioactive material encountered out of regulatory control. In a nuclear forensic examination, nuclear or other radioactive material is analysed using scientific instruments and analytical techniques capable of measuring the physical, chemical, elemental and isotopic properties of the material. A nuclear forensic examination results in the generation of technical data that support the needs and requirements of the investigative authority relating to the criminal circumstances of the event. A nuclear forensic examination may utilize one or several analytical techniques to examine the material, resulting in the generation of data that can be used in the investigation. While the instruments and analytical techniques themselves are an inherent aspect of nuclear forensic analysis, many States utilize subject matter experts with experience in using these instruments to perform the analyses. Subject matter expertise may aid the interpretation and reporting of nuclear forensic data to the investigative authority.

The development of an analytical plan for nuclear forensic examination is good practice for the generation of credible and reliable nuclear forensic data in support of investigations. The International Atomic Energy Agency (IAEA), in its Nuclear Security Series (NSS) No. 2-G (Rev. 1), Nuclear Forensics in Support of Investigations [1], provides guidance on the role of nuclear forensics in the context of investigating various types of criminal or other unauthorized act involving nuclear or other radioactive material. The publication also describes the development of a nuclear forensic capability, the role of nuclear forensics in a national nuclear security regime (including the investigation of criminal or other unauthorized acts involving nuclear or other radioactive material) and mechanisms for international cooperation and assistance in nuclear forensics.

1.2. OBJECTIVE

The purpose of this publication is to provide information for Member States on the use of nuclear forensics in response to criminal or other unauthorized acts involving nuclear or other radioactive material, based on the guidance provided in Ref. [1]. This publication is intended for technical and non-technical personnel responsible for developing and implementing nuclear

forensic analytical capabilities within the context of a broader national nuclear forensic programme stemming from national legislation.

1.3. SCOPE

This publication covers the conduct of a nuclear forensic examination in the context of commonly used techniques for analysing the physical, chemical, elemental and isotopic properties of nuclear or other radioactive material found out of regulatory control. It provides technical information that describes the application of specific methods, analytical techniques and technical expertise used for nuclear forensic analysis in support of investigations of criminal or other unauthorized acts involving nuclear or other radioactive material, the conduct of which may differ in each State. It also provides an understanding of the role of nuclear forensics within the broader investigation of criminal or other unauthorized acts involving nuclear or other radioactive material. Information contained in this publication can be adapted within the context of national legislative requirements relating to criminal investigations to the extent relevant to the needs of an individual State. Not all States may need the full extent of analytical instrumentation described in this publication.

1.4. STRUCTURE

Following this introductory section, Section 2 reviews how nuclear forensics supports investigations of criminal or other unauthorized acts involving nuclear or other radioactive material. Section 3 describes activities that may occur at a radiological crime scene, including categorization and initial characterization of nuclear or other radioactive material. Section 4 considers the handling of nuclear or other radioactive material upon receipt by a nuclear forensic laboratory. Section 5 addresses the characterization of nuclear or other radioactive material using non-destructive analytical (NDA) methods. Section 6 describes the characterization of nuclear or other radioactive material using non-destructive analytical (NDA) methods. Section 7 describes some quality assurance measures that may be taken during nuclear forensic examinations. Section 9 describes some mechanisms for international cooperation and assistance. Annexes I and II contain additional information regarding the handling of the types of nuclear or other radioactive material that may be the subject of a nuclear forensic examination, as well as detailed descriptions of analytical techniques that can be used during a nuclear forensic examination.

2. NUCLEAR FORENSICS IN SUPPORT OF INVESTIGATIONS OF CRIMINAL OR OTHER UNAUTHORIZED ACTS INVOLVING NUCLEAR OR OTHER RADIOACTIVE MATERIAL

A nuclear forensic examination, conducted in support of a criminal investigation related to an occurrence of nuclear or other radioactive material out of regulatory control, involves the use of analytical techniques to determine the physical, chemical, elemental and/or isotopic properties of the material. Data generated in the nuclear forensic examination can then be used, in consultation with subject matter experts, by the State investigative authority in furtherance of the investigation. The types of data needed by the investigative authority may vary significantly due to the wide variety of nuclear or other radioactive material in use around the world, as well as the different types of criminal or other unauthorized act involving nuclear or other radioactive material, specific circumstances of each case and differences in national laws.

Nuclear forensics is typically conducted in a laboratory setting by technical personnel with training and experience in the handling and analysis of nuclear or other radioactive material. During an investigation, the State investigative authority may ask a State nuclear forensic laboratory to perform a nuclear forensic examination in order to generate data on the physical, chemical, elemental and/or isotopic properties of the interdicted nuclear or other radioactive material, or of associated non-radioactive material. In some investigations, the use of relatively straightforward methods and instrumentation to identify and quantify the presence of specific radioactive isotopes in the material is sufficient to aid the investigation. However, in other investigations, additional, more sensitive and/or more precise analyses could be required to generate the types of data needed to support the investigation. In some cases, nuclear forensics can provide the data necessary for investigators to clearly identify the producer or last legal owner of the nuclear or other radioactive material, as well as its history outside of regulatory control. In other cases, nuclear forensics can be used to exclude possible origins or proliferation pathways.

Nuclear forensic analyses are typically performed at a laboratory that has been authorized by the State to receive and analyse nuclear or other radioactive material. This laboratory is often designated for nuclear forensic examination in the State's legal framework. If the legal framework has not yet been established in a State, it is possible for a State to task a nondesignated laboratory to conduct nuclear forensic analyses. However, the designation of a national nuclear forensic laboratory helps ensure that the designated laboratory is legally empowered to perform nuclear forensic analyses in a safe and secure environment, and that the data produced by the laboratory are authorized for use in the context of legal proceedings.

States may be able to use analytical techniques and subject matter expertise already present in the State to establish a nuclear forensic capability and conduct a nuclear forensic examination. The types of instrument useful in nuclear forensics, when operated by subject matter experts, can be found in nuclear research institutes, national or private laboratories, academic institutions, regulatory bodies, nuclear and radioactive material production plants, nuclear facilities, medical facilities, industrial facilities and other entities. If the investigation necessitates additional techniques that are not available in the State, international assistance from regional or international partners can be sought. For example, if a high-precision isotope ratio measurement capability is necessary to aid a specific investigation, and if the State does not currently have the capability to perform this type of measurement, other States may be able to assist.

When a criminal or other unauthorized act involving nuclear or other radioactive material occurs that results in an investigation that the State deems necessary for nuclear forensic analysis, many laboratories first develop a nuclear forensic analytical plan. Ref. [1] describes a model action plan for nuclear forensic analysis, which may be adapted for use by State nuclear forensic laboratories. Because the needs of the investigation may vary considerably due to the type of nuclear or other radioactive material under examination, many laboratories develop a unique analytical plan for each examination. Furthermore, the analytical plan may be adaptable in the event of the discovery of new information during the nuclear forensic examination.

A nuclear forensic examination often begins at the radiological crime scene or at the site of the discovery of the material if the event is not criminal in nature. Upon discovery, the nuclear or other radioactive material is often handled as evidence by many States, and chain of custody is maintained and documented from this point onward. Nuclear forensic analysis may also commence at this point, through in-field analysis of the physical, chemical, elemental and/or

isotopic properties of the material using field-deployable instruments operated by personnel trained on their use. Because of this, a State's nuclear forensic capability is intrinsically linked with its capability for managing crime scenes where nuclear or other radioactive material is known or suspected to be present. IAEA NSS No. 22-G, Radiological Crime Scene Management [2] provides guidance on radiological crime scene management.

A State's nuclear forensic capability is also linked with its ability to collect and analyse the types of traditional forensic evidence typically encountered in a criminal forensic examination, such as fingerprints, hair, fibres and many other types of evidence. The nuclear forensic model action plan describes the complementary role that nuclear forensics plays alongside traditional criminal forensics in the investigation [1]. If a State's nuclear forensic capability can function cooperatively and collaboratively with its criminal forensic capability, the State can ensure that all forms of nuclear and traditional forensic evidence are identified, retained and examined during the investigation.

Because each State has unique nuclear security needs and considerations, many States have chosen to develop nuclear forensic capabilities commensurate with their specific needs. A State may choose to possess a nuclear forensic capability sufficient to perform analyses of the types of nuclear or other radioactive material within regulatory control and current legal use by the State. For example, a State with a nuclear power plant or research reactor may possess the nuclear forensic capability to analyse pre- and post-irradiation nuclear fuels, as well as radioactive waste products and other radioactive material associated with reactor operation. Many States consider other factors as well, such as the history of smuggling and trafficking of specific types of nuclear or other radioactive material within or through the State, if known. For example, a State that has a history of interdicting nuclear or other radioactive material may develop a nuclear forensic capability that can analyse the specific types of material interdicted, even if the material does not originate from within the State's nuclear or other radioactive material holdings.

Confidence in nuclear forensic data can be demonstrated through a programme of quality assurance and quality control, achieved through laboratory accreditation, written and validated procedures, the employment of trained and experienced personnel, the use of standard reference materials and good practice in nuclear forensic analysis. Concise and clear technical reports resulting from nuclear forensic examination are ideal for addressing the needs and requirements of the investigative authority. Technical reports are used for establishing, clarifying and/or assessing data, facts and circumstances that are important for advancing investigations involving nuclear or other radioactive material. Data generated during the nuclear forensic examination any requirements by the State to retain evidence and supporting information that may be later requested by a court if nuclear forensic findings are used in legal proceedings.

Many States have worked to ensure that their nuclear forensic capabilities are sustainable, regularly exercised and are available to respond to a variety of possible nuclear security incidents. Sustainability can be ensured by using instrumentation and subject matter expertise that exists within the State. Although a State might have little experience in nuclear forensics in the early stages of establishing a nuclear forensic capability, for most analytical techniques, experience can be gained through workshops, trainings and exercises in nuclear forensics. Some States have organized internal nuclear forensic exercises and have participated in bilateral, regional or international exercises.

3. ACTIVITIES AT A RADIOLOGICAL CRIME SCENE INCLUDING CATEGORIZATION AND INITIAL CHARACTERIZATION OF NUCLEAR OR OTHER RADIOACTIVE MATERIAL

For many States, nuclear forensic examination begins at a radiological crime scene after the scene has been secured, all hazards have been identified and managed and all threats to life and property have been addressed. Decisions made and actions taken at a radiological crime scene can affect the conduct of a nuclear forensic examination, the quality of laboratory measurements and the legal defensibility of findings. In some cases, nuclear or other radioactive material out of regulatory control can be detected passively using radiation detection equipment (e.g. radiation portal monitors) installed at designated points of entry (e.g. border crossings, seaports, airports). In other cases, law enforcement personnel may detect the presence of radiation at a crime scene using handheld devices capable of informing the responding officers of the presence of gamma or neutron radiation above background levels. In the context of broader efforts to combat nuclear smuggling and trafficking, law enforcement personnel may also receive information from other sources regarding specific types of nuclear or other radioactive material out of regulatory control that might be in the possession of smugglers. In all cases, responding personnel typically utilize special procedures when dealing with radioactive hazards as they work to identify and secure nuclear or other radioactive material out of regulatory control.

Reference [2] describes radiological crime scene management as "the process used to ensure safe, secure, effective and efficient operations at a crime scene where nuclear or other radioactive material is known, or suspected, to be present." It describes in detail the framework and functional elements for managing a radiological crime scene, with particular emphasis on responding to radiological hazards and securing nuclear or other radioactive material for eventual examination by the nuclear forensic laboratory. The actions taken at the radiological crime scene by responding law enforcement personnel may impact, positively or negatively, the conduct of a nuclear forensic examination, the quality of the data generated and the confidence in conclusions based on the data. It is good practice to ensure that all responders to a criminal or other unauthorized act involving nuclear or other radioactive material remain aware of the needs of a subsequent nuclear forensic examination and understand how their actions can affect the legal defensibility of findings. Nuclear or other radioactive material out of regulatory control may constitute both traditional and nuclear forensic evidence, and it is good practice for the laboratory to ensure that the integrity of both types of evidence is maintained. Good communication between the investigative authority, in-field responders and the nuclear forensic laboratory can help to ensure that all evidence is handled and examined properly.

According to Ref. [1], "[c]ategorization is performed to identify nuclear security implications and the risk of the seized material to first responders, law enforcement personnel and the public". In contrast, "[c]haracterization is performed to determine the nature of the radioactive material and associated evidence" [1]. In the context of nuclear forensics, the physical, chemical, elemental and isotopic properties of nuclear or other radioactive material are often referred to collectively as nuclear forensic signatures.

Categorization measurements are often performed in the field, at the site of the interdiction or discovery of nuclear or other radioactive material out of regulatory control. Initial characterization measurements can also be performed in the field if the necessary field-deployable analytical instrumentation is available. Initial in-field characterization

measurements can provide preliminary data to assist the nuclear forensic laboratory in the development of an analytical plan. More precise, accurate and sensitive categorization measurements are usually performed when nuclear or other radioactive material is received at the nuclear forensic laboratory. Categorization and characterization measurements can also provide valuable information that may assist the investigative authority in maintaining the integrity of evidence, in preserving items of evidentiary value and in informing the State of the appropriate response to the criminal or other unauthorized act involving nuclear or other radioactive material.

In-field categorization measurements are often performed using handheld or field-deployable gamma and neutron analytical equipment by personnel familiar with their operation and use [3]. In some cases, law enforcement personnel work in partnership with gamma spectrometry experts and other relevant subject matter experts for in-field analysis. Some States choose to deploy gamma spectrometry experts to the field, under the guidance and supervision of law enforcement personnel, to conduct measurements. In other cases, measurements may be performed by law enforcement personnel or other responders trained in the operation and use of field-deployable gamma spectrometry equipment. In either case, the personnel performing in-field categorization and characterization measurements often remain in communication with laboratory-based subject matter experts, who can provide expert guidance on interpretation of gamma spectra and additional guidance on the handling of various types of nuclear or other radioactive material.

Low resolution handheld gamma spectrometers are useful for performing initial dose rate measurements and preliminary isotope identification and can also be used to locate additional nuclear or other radioactive material through secondary inspections of items at a radiological crime scene. These initial measurements can guide law enforcement personnel in the handling, collection, packaging and shielding of nuclear and other radioactive material, ensuring the continued safety of responding personnel and maintenance of evidence integrity.

More precise measurements can be performed using field-deployable high resolution gamma spectrometers. These instruments can reveal the presence and abundance of radionuclides that are challenging to identify using low resolution instruments. However, high resolution gamma spectra may be difficult to interpret without expert guidance, and in-field analysts may choose to consult gamma spectrometry experts and other relevant subject matter experts during and after data collection.

If gamma spectral data are consistent with the identification of nuclear material such as uranium or plutonium, in-field neutron detectors can be used for additional categorization measurements and can provide useful information regarding the hazard presented by the nuclear material. The Convention on the Physical Protection of Nuclear Material, as amended, provides a categorization of nuclear material into Categories I, II and III, in descending order of risk, and includes provisions on the physical protection of this material [4]. Radioactive sources can be categorized following the recommendations provided in IAEA Safety Standards Series No. RS-G-1.9, Categorization of Radioactive Sources [5], from Category 1 (high risk) to Category 5 (low risk).

Completion of in-field categorization and characterization measurements may result in a more complete understanding as to the nature of, and hazards presented by, the nuclear or other radioactive material. Categorization data may be shared with personnel responsible for material storage and transport, as well as other stakeholders involved in the investigation. In some cases,

in-field categorization and characterization measurements may be sufficient to address the needs and requirements of the investigative authority. However, in many cases, in-field measurements might not meet the evidentiary standards of the State and might lack the accuracy, precision or sensitivity of comparable measurements performed in a laboratory setting.

4. RECEIPT AND HANDLING OF SAMPLES AT THE NUCLEAR FORENSIC LABORATORY

The activities of the nuclear forensic laboratory may often follow the steps outlined below:

- 1. Receipt of a request for nuclear forensic analysis from the State investigative authority alongside relevant in-field categorization and initial characterization data;
- 2. Development of a nuclear forensic analytical plan for the specific type of nuclear or other radioactive material;
- 3. Receipt of material while maintaining chain of custody (this step is often performed in parallel with step 2);
- 4. Analysis of material upon receipt to confirm accuracy of in-field categorization;
- 5. Unpacking of the material, and documenting the status of the material upon receipt using photography and/or videography, alongside preservation of packaging, documentation and preservation of any associated traditional evidence;
- 6. If necessary, aliquoting (subsampling the material into smaller portions for analysis) for subsequent nuclear forensic examination.

4.1. RECEIPT OF REQUEST FOR NUCLEAR FORENSIC ANALYSIS FROM THE INVESTIGATIVE AUTHORITY ALONGSIDE RELEVANT IN-FIELD CATEGORIZATION AND CHARACTERIZATION DATA

A nuclear forensic examination is driven by the needs and requirements of the investigative authority. The role of the nuclear forensic laboratory is to support the investigation through the measurement of nuclear forensic signatures in nuclear or other radioactive material and associated non-radioactive material that is the subject of the investigation. At the start of the investigation, the investigative authority may communicate a preliminary list of questions to the laboratory, for which the laboratory can generate data to address. Some examples of the types of question the investigative authority might seek nuclear forensic data to answer include the following:

- What are the most abundant radionuclides present in the material?
- What is the total specific activity of the material?
- What is the isotopic composition of the material (for special nuclear material)?
- What is the category of radioactive source (for a radioactive source)?
- What is the origin of the material (or can any possible origins be ruled out)?
- Is the material consistent with a particular segment of the nuclear fuel cycle?
- What is the probable method of production in the nuclear fuel cycle?
- What pathway did the material take after loss from regulatory control?
- Are the physical, chemical, elemental and isotopic properties of the material consistent or inconsistent with nuclear or other radioactive material from a particular facility within the State?
- Is there more material like it out of regulatory control?

To assist in laboratory preparation for the arrival of the material, the investigative authority may share the results of in-field measurements, such as the dose rate and preliminary measurements of radioactive isotopes. The investigative authority may also provide any additional relevant information to assist the laboratory in the handling and analysis of the material, such as the presence of dispersible radioactive material, traditional forensic evidence that may need preservation and other considerations. This information can aid the laboratory in the development of an analytical plan and can help ensure that laboratory personnel are prepared to unpack and handle the material [6].

Communication between the laboratory and the investigative authority can be facilitated if a laboratory point of contact is established. The point of contact, often the manager of the nuclear forensic laboratory or another associated expert in nuclear forensics, is often the person with the most comprehensive understanding of the capabilities of the laboratory as well as legal considerations related to nuclear forensic examination. It is possible that a representative or representatives of the laboratory may be invited to provide expert witness testimony in legal proceedings related to the investigation. The point of contact therefore typically has the knowledge and experience required to explain key technical aspects of nuclear forensics to non-technical audiences.

4.2. DEVELOPMENT OF A NUCLEAR FORENSIC ANALYTICAL PLAN FOR THE SPECIFIC TYPE OF NUCLEAR OR OTHER RADIOACTIVE MATERIAL

A nuclear forensic analytical plan, developed through coordination and communication between the laboratory and the investigative authority, specifies the analytical technique or techniques to be performed in the nuclear forensic examination and describes the sequence in which they may be performed. The analytical plan may also describe which laboratory personnel will perform each measurement, as well as the quantity of the sample that is needed for measurement, how long each analysis will take and other considerations relevant to the examination. A nuclear forensic analytical plan is typically designed to ensure that nuclear forensic signature data that meet the needs and requirements of the investigative authority are generated during the examination. Development of the nuclear forensic analytical plan is typically also informed by the results of in-field categorization measurements and by the circumstances of the material interdiction or discovery.

Some laboratories aim to produce a series of analytical reports that may be provided to the investigative authority at intervals of 24 hours, one week and two months after receipt of the material by the laboratory. These timelines may be modified to meet the needs and requirements of the investigative authority. Clear understanding and communication of expected reporting timescales as part of the analytical plan development and modification process can help ensure that all parties remain aware of the expected progression of the nuclear forensic examination. Reporting timescales may account for not only the time needed for completion of measurements and data analysis, but also the time needed to compile, interpret and document the findings. Depending on the complexity of the nuclear forensic examination, the compilation and interpretation of data may take longer than the measurements themselves.

Once the investigative authority and the nuclear forensic laboratory concur on the analytical plan, laboratory analysis can commence. The analytical plan may be modified during the course of the examination for various reasons, such as in response to developments in the investigation or as a result of laboratory findings. For example, the discovery of the presence of a specific radionuclide may lead the investigative authority to request a more precise analysis of the

abundance of that radionuclide. Through regular reporting to the investigative authority during the course of the examination, all stakeholders can remain aware of developments that may necessitate adjustments to the analytical plan.

4.3. RECEIPT OF MATERIAL WHILE MAINTAINING CHAIN OF CUSTODY

It is good practice for the nuclear forensic laboratory to follow policies and procedures consistent with national evidentiary requirements for use in a criminal investigation to ensure that integrity of evidence is maintained throughout. Chain of custody is often initiated at the scene of nuclear or other radioactive material interdiction or discovery and maintained during transport and upon receipt by the laboratory. Many States use photography to document the initial status of nuclear or other radioactive material alongside packaging and shipping materials. It is good practice to ensure that chain of custody is also maintained within the nuclear forensic laboratory. A laboratory information management system can be used to track the material within the laboratory following similar procedures consistent with national laws and regulations on evidence handling within criminal forensic laboratories.

4.4. ANALYSIS OF MATERIAL UPON RECEIPT TO CONFIRM ACCURACY OF IN-FIELD CATEGORIZATION

Material unpacking is typically performed only after the nuclear forensic analytical plan has been developed, and after the material has been transferred to a secure, controlled area authorized for the handling of nuclear or other radioactive material. In most cases, laboratory personnel will have some information as to the nature of the material, as well as possible hazards presented by the unpacking and handling of the material, as a result of in-field categorization and characterization measurements. However, owing to the challenge of analysing nuclear or other radioactive material in suboptimal field conditions, categorization and characterization measurements performed in the field are often less precise and accurate than characterization measurements performed in the laboratory. Many laboratories choose to replicate in-field categorization measurements in the laboratory before and during the unpacking process [7]. Because of the radiation shielding effect caused by packaging materials, confirmation of categorization measurements might not be possible until some or all layers of packaging have been removed. Failure to perform these measurements may result in the development of an analytical plan that fails to call for the analysis of key nuclear forensic signatures and the exposure of laboratory personnel to unexpected radiation hazards. Of particular concern is the release of dispersible radioactive material within the workspace, which might contaminate the laboratory, compromise the integrity of ongoing work and injure laboratory personnel [8].

Prior to the arrival of the material, in order to ensure the safety of laboratory personnel and to minimize the possibility of cross contamination, the nuclear forensic laboratory may have already identified a workspace for the unpacking and initial handling of the material. An ideal workspace for unpacking and handling is one that has been cleaned thoroughly prior to the arrival of the material and is capable of receiving material with potentially high levels of radioactivity and/or dispersible radioactive material. Many nuclear forensic laboratories make use of a dedicated fume hood, which is cleaned and prepared for use in a nuclear forensic examination upon notification by the investigative authority of the imminent arrival of nuclear or other radioactive material is typically performed in the laboratory using engineering controls such as glove boxes or glove bags, as well as personal protective equipment such as laboratory-grade

single use gloves, laboratory coats, masks and safety glasses. Dosimetry equipment can be employed by laboratory personnel throughout the unpacking and handling process to continually assess the radiation hazard.

4.5. UNPACKING AND DOCUMENTATION OF MATERIAL, AND PRESERVATION AND DOCUMENTATION OF ANY ASSOCIATED TRADITIONAL FORENSIC EVIDENCE

It is good practice to thoroughly document the unpacking process, including the characteristics of the packaging material itself, as well as the key features and relationships of packaging layers separating different material. Photography and videography are useful tools for recording details that might be difficult to document in written notes. Many laboratories include a sample identification number in photographs and videos of each material, as well as some form of visual scale, such as a ruler and colour scale. When unpacking is complete, the laboratory may verify the received evidence inventory before proceeding, ensuring that all nuclear and radioactive material, and associated non-nuclear material have been received. Along with the nuclear or other radioactive material itself, the inventory could include packaging materials, traditional evidence, swipes taken to assess radioactive contamination, non-radioactive material associated with nuclear or other radioactive material and other items. Discrepancies between the expected and received material may be communicated to the investigative authority prior to proceeding with the nuclear forensic examination.

At this point of the nuclear forensic examination, the nuclear or other radioactive material under examination is typically treated by the laboratory as one or more nuclear forensic samples to be subjected to analysis, in whole or in part. Hereafter, this publication will use the terms 'sample' or 'samples' to refer to the nuclear or other radioactive material under nuclear forensic examination in the laboratory setting. Many laboratories assign a unique identifier to each sample at this stage of the examination.

During unpacking, the laboratory may identify traditional forensic evidence associated with the material. It is good practice to avoid alteration or destruction of traditional forensic evidence as much as possible. For example, swiping the samples to monitor for dispersible radioactive material may be performed in a location of the sample where fingerprints are least likely to be encountered. If traditional forensic evidence is encountered during the unpacking process, it can be documented for assessment by the investigative authority and assessed for appropriate follow-on analysis by a criminal forensic laboratory. If possible, it is good practice for unpacking to be supported by experts who can assist in identifying and preserving traditional forensic evidence.

Many States choose to designate for nuclear forensic examination a laboratory that routinely performs other analytical work involving nuclear or other radioactive material for other applications (e.g. environmental radioactivity measurements, characterization of nuclear fuels for the nuclear power industry or for analysis of post-irradiation research reactor targets). This can be challenging for laboratory personnel, as the needs of a nuclear forensic examination may be different than those of routine laboratory work. Many laboratories take steps to ensure that samples are not contaminated by other samples that may be present in the laboratory. Conversely, because nuclear forensic samples may be far more radioactive than the types of sample handled for routine laboratory analysis, many laboratories take additional steps to ensure that routine laboratory work is not compromised upon completion of nuclear forensic examination. Furthermore, nuclear forensic samples may contain radionuclides, such as

plutonium, with which the laboratory may be unfamiliar and/or not licensed to work. For these reasons, it is often beneficial for a State to identify and designate a nuclear forensic laboratory that is licensed to handle a wide variety of radioactive and nuclear material types.

4.6. ALIQUOTING FOR SUBSEQUENT NUCLEAR FORENSIC EXAMINATION, IF NECESSARY

Depending on the needs and requirements of the investigative authority, a nuclear forensic examination may involve the use of several analytical techniques including destructive analytical techniques, necessitating sample aliquoting. If aliquoting is necessary, consideration may be given to the physical form of the samples and the available amount of mass in each sample. The extent of physical, chemical, elemental and isotopic heterogeneity may also be considered, which can occur at the microscale and might not be visibly apparent. Characterization measurements may reveal the presence of heterogeneity that can be further examined in subsequent analyses, especially using spatially resolved techniques [9].

In bulk solid samples, heterogeneity may manifest as structural features or localized inclusions, and the aliquoting of the material may be performed in such a way as to reveal, emphasize or at least preserve such features, or to highlight bulk characteristics. In powder samples, it is possible that individual grains may be representative of one or more populations of grains. Grain populations may be characterized as distinctive in terms of physical, chemical, elemental and/or isotopic properties, and powder samples may contain two or more grain populations that can be distinguished and characterized using a variety of analytical techniques [10]. Alternatively, grain composition may vary over a wide range, representing a continuum of compositions [11]. Because of the inherent complexity of powder samples, laboratory personnel may attempt to characterize such samples at bulk and at single particle scales using spatially resolved analytical techniques, if available [12]. For slurries or liquid samples, it may be possible to separate solid or colloidal components from the bulk liquid, as the composition of these features may be distinct from the liquid component. In all cases, laboratory personnel can work to ensure that enough sample material remains after aliquoting so that all techniques called for in the analytical plan can be performed. If available, the remaining sample mass can be archived at the laboratory or kept in the custody of the investigative authority.

A more detailed review of sample handling and analysis considerations for nuclear or other radioactive material for nuclear forensic examination can be found in Annex I of this publication.

5. CHARACTERIZATION THROUGH NON-DESTRUCTIVE ANALYSIS

NDA is the analysis of a nuclear forensic sample in a manner that does not involve the physical alteration or consumption of the sample. For this reason, NDA methods are among the first analyses performed in a typical nuclear forensic examination. Depending on the needs and requirements of the investigative authority, NDA techniques alone might or might not be sufficient to generate nuclear forensic data to aid the investigation. In other cases, data obtained using NDA techniques can guide aliquoting and enable laboratory personnel to perform higher precision and more sensitive analyses using other analytical techniques.

A wide variety of NDA techniques may be used to perform measurements of the physical, chemical, elemental and/or isotopic properties of nuclear or other radioactive material. The selection and application of any given NDA technique is dependent on the needs and

requirements of the investigative authority, the nature of the samples and other considerations. NDA techniques are outlined in this section and discussed in more detail in Annex II of this publication. These techniques can provide data to assess the following:

- The physical characteristics of sample material;
- The identity and quantity of radionuclides present (including isotopic composition);
- The elemental composition;
- The chemical form (e.g. metal, oxide, alloy, ceramic, glass) and the identity and abundance of its crystalline constituents;
- The physical, chemical, elemental and isotopic heterogeneity at the micro- to macroscale.

For many NDA techniques, it may be possible to collect and report data to the investigative authority within 24 hours. For additional information on reporting timelines, which may be modified to fit the needs of the individual investigation and nuclear forensic analytical plan, see Ref. [1].

5.1. RADIOMETRIC NON-DESTRUCTIVE ANALYTICAL TECHNIQUES

Radiometric NDA techniques, especially gamma spectrometry techniques, are the foundation of a State's nuclear forensic capability. Because nuclear forensic examinations are primarily concerned with the radioactive properties of nuclear or other radioactive material out of regulatory control, gamma spectrometry is effective for rapid and precise measurements of these properties [13]. Gamma spectrometry can be performed using a range of instruments, from field-portable spectrometers capable of radioactive isotope identification at low energy resolution (e.g. scintillator detectors) to spectrometers capable of determining several radioactive isotopes present in a sample with high energy resolution (e.g. semiconductor detectors, such as high purity germanium detectors). When calibrated using standard reference materials, high resolution gamma spectrometry instruments are also capable of determining the specific activity of individual radioactive isotopes present in the sample (a measure of their overall abundance in the sample), as well as the isotopic composition of some elements with multiple gamma emitting isotopes.

Gamma spectrometry instruments, when operated by trained and experienced personnel, can provide a wealth of data to aid investigations. However, gamma spectral data can be complex to interpret, and misinterpretation can lead to erroneous conclusions. Gamma spectrometry specialists typically utilize spectral interpretation software to assist in interpretation of gamma spectrometry data, and most gamma spectrometry instrument manufacturers include or make available for purchase software for spectral analysis. Specialized gamma spectrometry software may be used for specific applications, such as the accurate determination of uranium and plutonium isotope ratios in nuclear material. Free, open-source software packages for gamma spectrum analysis are also available from several sources.

Commercially available gamma spectrometers vary in terms of resolution (broadly defined as low resolution, medium resolution and high resolution), cooling system (electrical or liquid nitrogen cooling), portability (field-portable or fixed) and other parameters. The choice of gamma spectrometer to use for a specific application in nuclear forensics depends on several variables, including the type of nuclear or other radioactive material, sample size and total specific activity. High resolution instruments, such as high purity germanium spectrometers used by many nuclear forensic laboratories around the world, are capable of resolving small differences in gamma radiation energies, enabling identification and quantification the radionuclides present with a higher degree of confidence [7]. Field-deployable high purity germanium spectrometers may also be used at a radiological crime scene for in-field characterization of material, enabling in-field radionuclide identification with a higher degree of confidence. However, high resolution gamma spectrometers may involve significantly longer analytical times than low resolution instruments, especially if samples are only weakly radioactive. Ease of use is also a consideration, as many field-portable low resolution gamma spectrometers are simpler to use, and spectral interpretation can be handled by the pre-installed software with minimal operator involvement [3]. However, confidence in radionuclide identification might be significantly lower than with high resolution instruments. In most cases, a sequential approach can be taken, beginning with initial low resolution analysis in the field using handheld, low resolution instruments, followed by confirmation using higher resolution instruments in the field or in the laboratory. Additional analyses using high resolution instruments with longer count times can confirm the detection of isotopes identified in the field with high confidence and may reveal the presence and abundance of other gamma emitting radionuclides [14]. High resolution instruments are useful for additional measurements, such as accurate and precise isotope ratio determination [15].

Gamma spectrometry has many specific applications in a nuclear forensic examination. For example, gamma spectrometry may be used to determine the presence and abundance of fission and activation products in an irradiated uranium fuel material [16–18], which may then be used alongside other data to determine the type of reactor (or rule out possible reactor types) in which the material was irradiated [19]. Gamma spectrometry may also be used to characterize the radioactive properties of industrial or medical radiation sources encountered out of regulatory control [20, 21]. In some cases, the age of relatively pure radioactive material can be determined by calculating its radiochronometric age (a practice known as radiochronometry), based on the abundance of daughter nuclides relative to parent nuclides [22, 23]. Gamma spectrometry is particularly suited to the determination of the age of plutonium, using the ²⁴¹Pu–²⁴¹Am parent–daughter system [24].

5.2. PHYSICAL NON-DESTRUCTIVE MEASUREMENTS

Characterization of the physical nuclear forensic signatures of nuclear forensic samples includes analysis of features such as physical form (e.g. solid, powder, liquid), colour, morphological features, surface textures, dimensions (e.g. length, width, thickness, height and dimensions of specific features), mass, density and distinctive physical features. Physical characterization also includes microscale features, such as grain size, grain shape and other morphological features measurable using microscopy techniques. Physical measurements typically do not need advanced instrumentation to be performed, and useful data can be collected using instrumentation such as an optical microscope and high resolution digital camera. In some cases, physical NDA measurements may be sufficient to address the needs and requirements of the investigative authority [25, 26]. In cases where sealed radioactive sources are the subject of nuclear forensic examination, labels, serial numbers or other unique identifiers may be present and may be of high diagnostic value for the determination of the material origin [27]. In other cases, images of distinctive physical features, such as imperfections or evidence of damage (e.g. scratches, chips), can provide insight into the history of the material after loss from regulatory control. These features may suggest, for example, the manner in which the material was handled and whether pieces or fragments of the material might still remain out of regulatory control [28].

Prior to other measurements, nuclear forensic samples can be photographed from a variety of angles and at various magnifications. Clear labelling of photographs ensures that each individual sample can be identified by an assigned unique identifier. Inclusion of a scale in each photograph ensures that dimensional data are captured at this stage of the examination. The colour of a nuclear forensic sample can be characterized using digital photography with certified colour charts, which may be purchased at low cost. It is useful to include some kind of colour reference because human perception of colour can vary significantly, and colour blindness is common in the general population. For more precise colour measurements, an ultraviolet/visible light spectrometer may be utilized. Colour is a diagnostic property of certain uranium-rich fuel cycle material, such as uranium ore concentrate, and can be indicative of the elemental and chemical composition of the material [29].

Mass is an inherently useful property of the nuclear or other radioactive material under nuclear forensic examination, and a precise and accurate determination of material mass is likely to be required by the investigative authority in most nuclear forensic examinations [28]. Mass can be determined with high accuracy and precision using a laboratory-grade analytical balance. Sample mass data are necessary for determining sample density, as well as for accurate calculation of elemental concentration following measurement using other instruments. If the sample is aliquoted, the mass of each aliquot is typically measured prior to subsequent analyses. Before determination of sample mass, the accuracy of the analytical balance can be assessed using calibration weights, which are commercially available from analytical balance manufacturers. Some sample matrices, such as plastic materials, may be susceptible to electrostatic effects that can result in erroneous laboratory balance measurements. In this case, anti-static devices may be employed to minimize this effect during the weighing process. The masses of liquids, slurries and other non-solid samples can be determined by massing within clean, pre-weighed beakers. Powder samples can be massed within pre-weighed beakers, or by using weighing paper, while taking precautions to ensure that dispersible radioactive material is not released during measurement.

The size of nuclear forensic samples may vary widely, from individual microscale radioactive particles present on the surface of another material to bulk nuclear or other radioactive material, such as uranium fuel pellets or bulk uranium metals of centimetre scale or larger. Sample size and shape (hereafter referred to as 'sample geometry') can be measured using an optical microscope and digital camera with a ruler, calliper or other measuring equipment. For more precise measurements, a three-dimensional (3D) scanner may be used to reconstruct the sample as a 3D model. Sample geometry data are particularly useful for investigations of uranium fuel pellets out of regulatory control because pellets are manufactured to meet the narrow specifications of specific reactor types [30]. Therefore, their masses and geometries can be diagnostic of the type of nuclear reactor for which they were intended, and these data may be used by the investigative authority to determine, or exclude, the identity of the producer or the last legal owner of the material [31]. Furthermore, sample mass and sample geometry data may be used to inform other analyses, such as the calibration of the full energy peak efficiency of a high purity germanium gamma spectrometer, allowing for the calculation of more precise radionuclide abundances and isotope ratios [7].

If the mass of a sample is known, its density can be calculated following a measurement of its volume. In the case of a solid sample with a simple 3D shape, such as a cylinder, its volume can be calculated using measurements of key dimensions, such as height and radius. A 3D scanner may be used to determine the volume of irregularly shaped samples. Alternatively, a gas or liquid pycnometer may be employed to determine sample volume using the displacement

method [28]. Two advantages of gas pycnometry over liquid pycnometry are its applicability to samples containing significant pore space, such as powders, and its lower likelihood of sample contamination. Density measurements can be useful as diagnostic nuclear forensic signatures of metallic alloys and ceramics used in the nuclear fuel cycle, among other applications.

Because optical microscopes are only capable of magnification up to ~1500×, techniques such as scanning electron microscopy (SEM) are useful for obtaining physical data at higher magnifications. Commercially available SEM instruments are capable of magnification up to ~100 000×, revealing physical features as small as 10^{-9} m. Utilizing SEM, it is possible to make qualitative observations and quantitative measurements (such as the physical dimensions of individual grains) on the microscale [19, 32]. Most SEM instruments have two modes, enabling the collection of high resolution two-dimensional (2D) images of micro- to nanoscale structural sample features and the determination of elemental composition, also at the micro- to nanoscale [33]. SEM is especially useful for assessing sample heterogeneity, which can guide additional aliquoting and analysis [34].

While SEM is nominally non-destructive, in most cases, the sample is prepared for analysis in some respect [35]. Most samples need mounting in the SEM sample chamber, which is often performed using electrically conductive tape which may contaminate the surface of the sample with metallic elements. As SEM measurements are performed at high vacuum, loose radioactive powders can be dispersed within the sample chamber, contaminating and potentially damaging the instrument as well as jeopardizing the safety of laboratory personnel. Accurate elemental measurements are usually obtained on a flat sample surface, which can be achieved by polishing prior to analysis but results in the destruction of sample surface morphology. In most cases, physical characterization using standard photographic and optical microscopic techniques is completed before SEM analysis is performed.

As some physical characteristics are qualitative in nature (e.g. involving physical descriptions of the material and its features that cannot be directly quantified), interpretation of physical nuclear forensic signatures is aided by subject matter experts [36]. Many States utilize laboratory personnel familiar with the types of nuclear or other radioactive material in their material holdings and their physical, chemical, elemental and isotopic properties. Inclusion of these experienced and knowledgeable personnel can significantly improve the interpretation of physical nuclear forensic signature data.

5.3. ELEMENTAL AND CHEMICAL NON-DESTRUCTIVE MEASUREMENTS

Gamma emitting radionuclides present in nuclear or other radioactive material may be measured using gamma spectrometry techniques. The remaining elements can be measured using other analytical techniques. In many types of nuclear or other radioactive material, the bulk of the sample mass consists of non-radioactive isotopes, and the presence and abundance of these elements may be measured using a variety of techniques. The most commonly used technique for non-destructive elemental analysis is X ray fluorescence spectroscopy (XRF), which can be used to measure the concentration of most elements that might be present in the sample. Because an XRF analysis can be performed in minutes, it is often the first technique used for elemental measurements in a nuclear forensic examination [37]. Most XRF instruments can be run without calibration for a specific nuclear or radioactive material type, instead relying on the fundamental parameters of X ray generation to enable calculation of the elemental concentration [38]. However, irregularly shaped samples and coarse-grained powder

samples benefit from geometric correction or the use of standard reference materials to ensure accurate measurement. Furthermore, some sample matrices present unique challenges for XRF. For example, the uranium in uranium-rich samples may generate spectral interferences via XRF that prohibit the determination of certain elements, such as molybdenum and zirconium.

Many SEM instruments are equipped with an XRF capability, enabling spatially resolved elemental analysis of nuclear forensic samples on the microscale (this capability is sometimes referred to as 'energy dispersive spectroscopy'). Using a SEM equipped with an energy dispersive spectroscopy capability, it is possible to measure the elemental composition of 'points' on the surface of a sample (on the scale of 10⁻⁶ m) and to generate elemental 'maps' of the surface of a sample, revealing elemental heterogeneity that might not be visible using optical microscopy [39]. Elemental heterogeneity may be correlated with physical heterogeneity also measured using SEM. This powerful approach has many applications for nuclear forensics. For example, it is possible to observe the locations of uranium within a glass sample or differences in elemental composition between various grains and grain populations in a powder sample [40].

X ray diffraction spectroscopy (XRD) is an NDA technique used for determination of the chemical composition (arrangement of atoms in crystalline or polycrystalline materials such as ores, metals, ceramics and oxides) of a nuclear forensic sample [41]. Nuclear or other radioactive material can consist of one or a mixture of several crystalline solids (sometimes referred to as 'phases') or non-crystalline constituents (such as glass), and XRD can measure the presence of, and in some cases the abundance of, the crystalline components [42]. Although the XRD analysis is itself non-destructive, the sample is typically loaded into the instrument in such a manner as to ensure a flat surface for analysis [43]. Therefore, this technique is ideally suited for analysis of fine-grained powder samples, which can be loaded in the spectrometer to produce a flat surface. Other types of sample can be analysed using XRD, but may involve some alteration, such as the crushing of a sample aliquot to a fine-grained powder prior to loading. Due to the inherently disordered arrangement of atoms in glass and other non-crystalline materials, XRD is incapable of quantifying these constituents.

6. CHARACTERIZATION THROUGH DESTRUCTIVE ANALYSIS

In some nuclear forensic examinations, NDA techniques might not be sufficient to generate the data necessary to support the needs and requirements of the investigative authority. In these cases, characterization through destructive analysis may be necessary. Destructive analysis techniques are often capable of providing more precise data on the physical, chemical, elemental and isotopic properties of nuclear or other radioactive material than NDA techniques. Furthermore, destructive analysis techniques tend to be more sensitive than NDA techniques. However, destructive analysis techniques often take longer to perform than NDA techniques and need greater expertise in the instrumentation, which is usually more expensive to procure and operate. Moreover, destructive analysis techniques necessitate sample aliquoting and sample preparation prior to analysis, and usually result in the consumption of the aliquot during analysis.

A wide variety of destructive analysis techniques may be used to perform measurements of the physical, chemical, elemental and isotopic properties of nuclear or other radioactive material. These techniques are discussed in more detail in Annex II. Selection and application of a specific destructive analysis technique or techniques typically depends on the needs and

requirements of the investigative authority. Several considerations are important when deciding which destructive analysis technique to use, if any. These include the following:

- The quantity and type of sample available and needed for the measurement;
- The sample preparation needs for the measurement;
- The amount of sample to be consumed during the course of the measurement;
- The possibility of heterogeneity in the physical, chemical, elemental and/or isotopic composition of the sample, which may result in the analysis of a non-representative aliquot.

Because the use of destructive analysis techniques necessitates consumption of a sample aliquot, it is impossible to replicate the measurement on a given aliquot. The production of multiple aliquots, and, in some cases, the transfer and reuse of aliquots from NDA techniques to destructive analysis techniques, is a good strategy for maximizing the data generated from a minimum amount of available sample. These considerations are particularly important when only a limited amount of sample mass is available for analysis.

6.1. ELEMENTAL AND CHEMICAL COMPOSITION MEASUREMENTS

An elemental composition dataset consists of the concentration of a set of elements that constitute the mass of the sample, along with corresponding measurement uncertainty for each element. A wide variety of analytical techniques can be used to measure elemental composition, though not all techniques are well suited for every type of nuclear or other radioactive material or for every element. For example, mass spectrometry techniques are ideal for metallic and semi-metallic elements but are poorly suited to non-metallic elements such as fluorine or chlorine. XRF can be used to quantify virtually every element that might be present in a sample, but it might lack the measurement precision, accuracy and sensitivity (especially for the lightest elements) to meet the needs and requirements of the investigative authority. Single element quantification methods, such as titration methods, can provide high precision elemental concentration data for one or a few important element(s), but might involve consumption of a significant amount of sample. The choice of technique is ultimately driven by factors such as the needs and requirements of the investigative authority, the type of sample under examination and the amount of sample available.

Mass spectrometry is a destructive analysis technique commonly used in nuclear forensics for determination of the elemental composition of many types of nuclear or other radioactive material. Although the field of mass spectrometry encompasses a wide range of techniques, including those used for the analysis of organic compounds, applications in nuclear forensics generally focus on quantification of the metallic and semi-metallic elements. Because of this, inductively coupled plasma mass spectrometry (ICP-MS) has gained widespread popularity in nuclear forensics [44]. This technique, which typically involves the dissolution of an aliquot of the sample in an acid matrix, makes use of a high temperature plasma to atomize and ionize the elements present within the sample. Ions are then accelerated under vacuum in a magnetic field, forming a beam of ions. The ions are separated on the basis of their mass-to-charge ratio using a mass analyser. The intensities of the separated ion beams are measured by one or more detectors within the instrument.

For elemental concentration measurements in a nuclear forensic examination, the use of a mass spectrometer with a quadrupole mass analyser is often useful. This type of mass spectrometer can accurately and precisely measure the metallic and semi-metallic elements in a sample within minutes to hours, excluding sample preparation time. Quadrupole ICP-MS instruments can determine the concentration of most metallic elements with a sensitivity of four to six orders of magnitude higher than XRF [45]. However, quadrupole ICP-MS involves optimization and calibration of each set of elements to be measured, and a new calibration is typically performed once or more per day. In addition, sample preparation and dissolution for mass spectrometry usually involves laboratory personnel working within a clean workspace or laboratory. Moreover, for this method, purified acids and single use disposable laboratory equipment are useful to ensure that the background for the measurements is kept as low as possible. The establishment of liquid radioactive waste streams ensures that waste generated during measurement is collected and properly disposed of. For these reasons, quadrupole ICP-MS necessitates considerable training and experience on the part of laboratory personnel, as well as significant laboratory infrastructure.

Inductively coupled plasma atomic emission spectroscopy (ICP-AES) is an alternative method for elemental composition measurements, with some specific applications to nuclear forensics [46]. ICP-AES functions similarly to ICP-MS, but instead of measuring the ion beams generated during atomization and ionization in the plasma, the instrument measures the characteristic electromagnetic radiation (usually around the visible light spectrum) produced during sample ionization. Although ICP-AES is less sensitive than ICP-MS for most of the elements that both can analyse, it can accurately and precisely measure the concentration of many elements that may of be of interest to the investigative authority. However, ICP-AES retains many of the limitations of ICP-MS, including the need for a clean preparation space and a radioactive waste stream.

XRF can also be performed as a destructive analysis technique. Powder samples can be converted into pressed pellets using a pellet press and binding material to minimize pore space and increase particle density, resulting in more accurate and precise concentration measurements [47]. Alternatively, solid samples can be melted using a furnace and fluxing compound, then quenched to produce a glass disc or bead, which can then be analysed by XRF. This approach minimizes sample heterogeneity effects by ensuring an even distribution of atoms within a glass matrix, resulting in more accurate elemental concentration data [48]. However, the production of glass discs and beads consumes a large amount of the sample, and the quality of the resultant glass is somewhat dependent on the composition of the sample and the speed at which the melt is quenched. Moreover, the melting and quenching of radioactive samples might present a dispersible radioactive material hazard.

There are several titration methods that can be used to determine single element concentrations in nuclear forensic samples [49, 50]. One titration method frequently used in nuclear forensics is the modified Davies-Gray technique, which is capable of accurate and precise measurements of uranium concentration [51]. This technique is ideally suited to uranium-rich material, such as uranium ore concentrate and uranium fuel pellets. Although this technique can be performed manually, a more reliable and reproducible approach is to utilize an auto-titration system, which can analyse eight or more samples in sequence, with reliability superior to that achievable by a human operator [52]. Although the application of this technique necessitates experience in benchtop chemistry, it is significantly less expensive than mass spectrometry, while still capable of producing high precision uranium concentration data comparable to some mass spectrometry techniques [53].

There are two primary options for determining the presence and abundance of chemical compounds using destructive analysis: XRD and Fourier transform infrared spectrometry

(FTIR). Preparation for XRD can involve the aliquoting and crushing of the sample into a finegrained powder prior to analysis, and this transformation of the aliquot can be considered destruction of the sample. This aliquot can then be used for other non-destructive or destructive analytical techniques, since the XRD measurement itself does not consume the sample [54]. Similarly, FTIR is in principle an NDA technique, but in most cases, the technique involves the grinding of an aliquot into a powder, followed by the formation of a pellet using a pellet press and binding material. The FTIR technique involves bombardment of the sample with infrared photons, resulting in the production of spectra that are characteristic of the molecular compounds present in the sample [55, 56]. The spectra can then be compared against the spectral patterns of known chemical compounds, allowing for identification of the compounds present.

Elemental and chemical composition data have many uses in investigations of nuclear or other radioactive material out of regulatory control, and the involvement of subject matter experts can aid interpretation [57]. In some cases, a suite of elemental data may be useful for determining the geological origin of the material. Many uranium ore and uranium ore concentrate materials have distinctive elemental profiles [58-64], in particular the lanthanide elements, which can be indicative of the type of ore deposit from which the material originated [65]. Linking these materials to geological and geographic origins may provide insight into the last legal owner of the material and may aid determination of where it was lost from regulatory control [66]. High concentrations of anionic groups such as fluoride, chloride, sulphate and phosphate are often found in trace levels in uranium-rich fuel cycle material originating from the use of mineral acids and other reagents during uranium extraction and purification [67, 68]. For instance, sulphuric and nitric acids are commonly used for the leaching and dissolution of uranium ore [69]. Although fluoride is often used in the nuclear fuel cycle for the storage of purified uranium-rich products (uranium tetrafluoride) or in enrichment processes (uranium hexafluoride), fluorine is not usually added intentionally during the production of uranium ore concentrate. For this reason, the likely sources of fluoride anionic groups in uranium ore concentrate are uranium bearing minerals, the presence of which can aid determination of geological origin [70]. Similarly, in cases where the uranium is recovered as a by-product from phosphate-based ores, an elevated level of phosphate in the uranium product may be expected.

Many other elements are indicative of nuclear fuel cycle processes. For example, the presence of elevated boron or gadolinium concentrations in a uranium fuel pellet may indicate that the addition of these elements was intentional during manufacture, as a neutron poison, and may be indicative of the type of nuclear reactor for which it was intended [71]. In another example, some fuel manufacturers use aluminium stearate as a mould release agent, which can result in elevated aluminium concentration in fuel pellets [72]. Contaminants on the surface of fuel pellets may also be present, which may provide information about pellet fabrication techniques used in production, such as sintering and grinding [52]. For example, trace amounts of molybdenum may be deposited on pellets from molybdenum trays used to hold the pellets during the sintering process.

6.2. ISOTOPIC COMPOSITION MEASUREMENTS

The ability to measure the isotopic composition of uranium and plutonium is useful for States in possession of nuclear material such as uranium fuel used in nuclear power reactors or research reactors. Because it is illegal in most States to possess enriched uranium or plutonium out of regulatory control, the investigative authority may request data on the isotopic composition of uranium or plutonium if uranium- or plutonium-rich materials are encountered out of regulatory control. Uranium and plutonium isotopic composition data are usually reported as isotope ratios, with the most abundant (major) isotope in the denominator. In many cases, the investigative authority is interested in ²³⁵U/²³⁸U and ²⁴⁰Pu/²³⁹Pu isotope ratios for uranium and plutonium, respectively.

The isotopic composition of uranium is highly variable in nuclear or other radioactive material. Natural uranium consists of ~0.725 atom % ²³⁵U (in ratio form, ²³⁵U/²³⁸U = 7.25 × 10⁻³ [73]). The abundance of ²³⁵U relative to ²³⁸U is effectively constant in natural materials containing uranium [74]. Anthropogenic enrichment of uranium and production of depleted uranium as a waste product of enrichment alter the uranium isotopic composition of nuclear fuel cycle material. Nuclear power plants normally utilize uranium fuels in the range of 0.725% (natural) to 6.5% ²³⁵U. Research reactors may utilize a much wider range of enrichment, from less than 20% to more than 90% ²³⁵U, although the use of high enriched uranium for civilian purposes has greatly declined in recent decades. All uranium-rich material contains ²³⁴U (natural ²³⁴U/²³⁸U \approx 5 × 10⁻⁵, with small variations in the naturally occurring ratio due to natural processes [75]). Although ²³⁵U and ²³³U do not occur naturally, they may be present in material that has been irradiated in a nuclear reactor, contaminated with irradiated material or exposed to neutrons in some other setting [76].

Plutonium is a non-naturally occurring element produced primarily through the irradiation of uranium in nuclear reactors. Nuclear or other radioactive material samples containing plutonium may contain the isotopes ²³⁸Pu, ²³⁹Pu, ²⁴⁰Pu, ²⁴¹Pu, ²⁴²Pu and ²⁴⁴Pu, and the relative abundance of these isotopes may vary significantly. Nuclear material can contain both uranium and plutonium, and some fresh nuclear power plant fuels consist of a mixture of both elements (mixed oxide fuels). Because plutonium is highly toxic and acutely radioactive, it is good practice for laboratory personnel to take extraordinary care when handling and analysing material containing plutonium during a nuclear forensic examination.

Although gamma spectrometry can be used to measure uranium and plutonium isotopic composition, this method is not as precise as many destructive analysis techniques, and ²³⁶U cannot be measured by gamma spectrometry. Mass spectrometry is a useful technique for precise analysis of the concentration and isotopic composition of uranium and plutonium. Several plasma mass spectrometry instruments are commonly used for these measurements, including, with increasing measurement precision: (i) quadrupole ICP-MS [77, 78]; (ii) single collector sector field ICP-MS [16, 79-81]; and (iii) multi-collector sector field ICP-MS [82, 83]. In addition, thermal ionization mass spectrometry (TIMS), which utilizes a heated filament instead of a plasma to atomize and ionize the sample for analysis, can produce uranium and plutonium isotopic composition data equal or superior to that of the most precise plasma source mass spectrometry instruments [24, 84-89]. Advanced mass spectrometry techniques necessitate significant technical expertise in isotope ratio analysis and have high operational costs, including the need for clean laboratory space and a continuous stable source of electricity to maintain vacuum systems. Furthermore, these methods involve the acquisition of expensive standard reference materials for calibration and quality control, which can incur additional costs.

Alpha spectrometry is a destructive analysis technique used to determine the specific activity and isotopic composition of alpha emitting radionuclides, such as those of uranium and plutonium [16, 90]. In a nuclear forensic examination, alpha spectrometry can be used to determine enrichment of uranium (²³⁵U/²³⁸U), grade of plutonium (²⁴⁰Pu/²³⁹Pu) and abundance of other alpha emitting radionuclides, such as ²³⁷Np, ²⁴¹Am, ²²⁶Ra [91]. In addition, alpha

spectrometry is an ideal instrumental technique for ²³⁸Pu analysis [92]. Alpha spectrometry is a cost effective alternative to advanced mass spectrometry techniques for many States. However, the use of alpha spectrometry involves chemical separation and purification of uranium and plutonium prior to analysis. Therefore, in addition to the alpha spectrometry system itself, a chemical separation capability is usually paired with alpha spectrometry for use in nuclear forensics. Chemical separation involves considerable technical ability and experience on the part of the analyst. Some environmental radioactivity laboratories are equipped with this capability for routine analysis, and States can adapt this capability for use in nuclear forensics. However, nuclear forensic samples may be considerably more radioactive than environmental samples routinely analysed by environmental laboratories, and it is possible to oversaturate an alpha spectrometry system in such a manner as to render it useless for low activity analyses. Dilution of the sample after chemical separation and prior to loading for alpha spectrometry analysis can help avoid alpha detector saturation and damage.

Gamma spectrometry, mass spectrometry and alpha spectrometry techniques are all capable of accurate and precise uranium and plutonium isotopic composition measurements. For example, all three sets of techniques can often determine the isotopic composition of low enriched and high enriched uranium with sufficiently high precision to determine if an interdicted nuclear material is enriched above the natural isotopic composition of 0.725% ²³⁵U. In this example, more precise analysis of uranium isotopic composition may be useful only if the material is close to natural isotopic composition or if the investigative authority requests more precise data for some other purpose. In these cases, it may be useful to utilize the more precise mass spectrometry techniques, or to seek assistance from other States if this capability does not exist within the State itself.

Interpretation of uranium and plutonium isotopic composition can be complex but may have considerable value in many investigations of nuclear material out of regulatory control. For example, ²³⁴U is inherently enriched alongside ²³⁵U in uranium enrichment processes, and therefore tends to be positively correlated with ²³⁵U in enriched uranium samples [93]. The ²³⁵U/²³⁸U and ²³⁴U/²³⁸U ratios of enriched uranium samples may be indicative of the type of enrichment technique used to produce the material [94]. In reprocessed uranium samples, the presence of ²³⁶U may indicate that the material was exposed to neutrons (irradiated) at some point in its history [95]. The isotopic composition of plutonium, especially the minor isotopes ²³⁸Pu, ²⁴¹Pu, ²⁴²Pu and ²⁴⁴Pu, may provide information about the type of reactor in which the plutonium was produced [96–100]. Different reactor types, such as light water moderated reactors, heavy water moderated reactors and graphite moderated reactors, have different neutron energy spectra and fluxes, as well as different initial ²³⁵U enrichments of the fuel, resulting in plutonium isotopic compositions that can be diagnostic of the nuclear reactor in which the material was irradiated [101–103].

6.3. RADIOCHRONOMETRY MEASUREMENTS

The accumulation of daughter nuclides in nuclear or other radioactive material occurs as parent isotopes decay over time. The constant rate of radionuclide decay can allow for calculation of the age of the material (termed 'radiochronometry', or sometimes 'age dating') if the daughter-to-parent nuclide ratio can be determined [104]. For example, the age of a uranium-rich material can be calculated using a precise measurement of the ratio of the parent isotope ²³⁴U to its daughter isotope ²³⁰Th [105]. The measured age of nuclear or other radioactive material may reflect the time when the separation of the daughter nuclide from its parent nuclide last took place. In this case, the age may also be referred to as the production date of the material

[90]. A crucial assumption for interpretation of the measured age as a production age is complete separation between parent and daughter nuclides at the date of production. Furthermore, for an age result to be accurate, it is assumed that the material behaves as a 'closed system'; that is, neither parent nor daughter nuclide has been removed or added during the time since production other than through decay of the parent nuclide and ingrowth of the daughter nuclide [106].

For daughter-to-parent isotope ratio measurements, precise mass spectrometry techniques are typically preferred over gamma spectrometry due to their superior precision. However, for plutonium-rich material, the age can be obtained using the ²⁴¹Am/²⁴¹Pu ratio by gamma spectrometry [24]. Alpha spectrometry can also be used for age determination of uranium and plutonium in some cases [90]. For improved confidence in the results obtained by age dating, at least two independent parent-to-daughter ratios can be used, as some production processes do not result in quantitative removal of all daughter isotopes from the nuclear or other radioactive material. Some examples of daughter-to-parent ratios used for age determination in nuclear forensics are ²³⁰Th-²³⁴U [107–110], ²³¹Pa-²³⁵U [111–117], ²⁴¹Am-²⁴¹Pu [24, 118], ²³⁵U-²³⁹Pu [119–121], and ²²⁷Ac-²³⁵U [122, 123].

In a nuclear forensic examination, determination of material production age can aid the investigation in several ways. Material age data can assist the investigative authority in determining, or ruling out, the facility from which the material was produced. States with a nuclear fuel cycle have records of when their facilities were producing nuclear or other radioactive material, enabling comparison between the material radiochronometry age and the known timescales of production. Material ages that fall outside of a State's history of nuclear or other radioactive material production may be indicative of material origin outside the State. Confidence in interpretation of material age is increased by analysing more than one parent–daughter pair, such as $^{234}U^{-230}Th$ and $^{235}U^{-231}Pa$ in uranium-rich material [124]. Age agreement between two or more parent–daughter pairs is strongly suggestive that the measured ages reflect the true age of material production [125]. Disagreement between measured ages suggests that one or both ages is inaccurate, possibly due to incomplete removal of the daughter nuclide during material production.

6.4. OTHER DESTRUCTIVE ANALYSIS TECHNIQUES

Some States may possess analytical techniques capable of measuring other nuclear forensic signatures not mentioned above. For example, it is possible to use TIMS or multi-collector ICP-MS to measure the strontium, neodymium, lead, molybdenum, samarium and thorium isotopic composition of uranium ore and uranium ore concentrate materials, which can enable linkage of these materials to specific geologic and geographic origins [126–132]. The presence and abundance of fission and activation products, such as ⁹⁰Sr, can also be measured with high precision in spent fuel and other material exposed to neutrons using these and other advanced techniques [133–138]. Novel isotopic systems, such as those employing molybdenum, can be used to determine (or rule out) the use of specific enrichment and other nuclear fuel cycle processes in the production of the material [139]. Using other types of mass spectrometer, such as dual inlet gas source instruments designed for light stable isotope measurements, it is possible to determine the isotopic composition of oxygen, hydrogen, carbon and other stable isotopes that are present in a material [140–144].

Advanced mass spectrometry techniques can be used to obtain high precision elemental and isotopic data from nuclear or other radioactive material on a spatial scale from tens to hundreds

of microns. Secondary ion mass spectrometry (SIMS) is a spatially resolved analytical technique capable of determining the uranium isotopic composition of nuclear forensic samples on the microscale [145, 146]. This technique has been used to measure microscale uranium isotopic heterogeneities in uranium fuel pellets and to measure the isotopic composition of individual nuclear material particles. Approaches taken by the nuclear safeguards community using SIMS can be applied to particulate nuclear forensic samples [147]. The use of multicollector sector field ICP-MS instruments in tandem with laser ablation sample introduction systems has demonstrated that this approach can reveal isotopic heterogeneity on the scale of 1×10^{-6} m in bulk and particulate nuclear material [9, 12, 79, 148, 149]. One additional advantage of laser ablation sample introduction systems is the minimal consumption of sample mass during analysis.

7. QUALITY ASSURANCE IN NUCLEAR FORENSIC ANALYSIS

For the results of a nuclear forensic examination to be useful to the investigative authority, the nuclear forensic laboratory often demonstrates and communicates the quality and reliability of measurements in a variety of ways. The nuclear forensic laboratory can take several measures to ensure the quality of nuclear forensic analyses. Many nuclear forensic laboratories have in place a quality assurance and quality control programme. Since most designated nuclear forensic laboratories perform analytical work other than nuclear forensics, they may already be operating under some form of quality assurance and quality control programme for routine work. Many laboratories are certified to meet the International Organization for Standardization (ISO) international standard, which demands the demonstration of reliable results in routine audits. Although a nuclear forensic laboratory is not required by any international body to operate under an international standard such as ISO 17025 (although they may be required to operate under this standard by national law), accreditation increases confidence in the analytical results produced by the laboratory [150].

Most nuclear forensic laboratories follow standard operating procedures for the analyses they are asked to perform in a nuclear forensic examination. Very often, these standard operating procedures are adapted from those used for routine laboratory analysis, such as the analysis of solid radioactive material by gamma spectrometry. Whenever possible, the nuclear forensic laboratory can develop standard operating procedures for the analytical instruments and types of nuclear or other radioactive material they might expect to analyse in the context of nuclear forensics. However, it is helpful for laboratory personnel to remain flexible, since it is possible to receive unexpected types of material for which the laboratory does not have a standard operating procedure prepared in advance. Deviations from a standard operating procedure can be documented in order to ensure that measurement quality is maintained and demonstrated to the investigative authority.

Analyses of standard reference materials alongside nuclear forensic samples are essential for ensuring measurement accuracy [151]. Standard reference materials are typically certified for one or more analytes of interest, in a relevant matrix. For example, a uranium ore concentrate standard reference material may be certified for iron concentration, among other elements, as well as for uranium isotopic composition. In a nuclear forensic examination, most laboratories measure one or more standard reference materials alongside the nuclear forensic sample [152], ensuring that each measured analyte is associated with a corresponding standard reference material measurement. Before reporting data, laboratory personnel typically conduct assessments of standard reference material measurement results in order to ensure that results are within certified values including uncertainties. In addition, it is good practice to use standard reference materials that match the matrix of the nuclear forensic sample as closely as possible [153]. For example, a uranium fuel pellet standard reference material can be analysed alongside the examination of a uranium fuel pellet material out of regulatory control [154–157]. Standard reference materials for elemental and isotopic composition may be purchased from several suppliers, and new reference materials are continually in development [158–160], including for new analytes [161] and for spatially resolved analysis [162].

Because the need for nuclear forensic examinations may occur infrequently in a State, laboratory personnel may take the opportunity to exercise and improve their skills in nuclear forensic analysis whenever possible. The international community of nuclear forensics practitioners routinely participates in bilateral and multilateral opportunities to exercise the process of a nuclear forensic examination. The 'Collaborative Material Exercise' series of exercises, organized by the Nuclear Forensics International Technical Working Group, is an example of the type of nuclear forensics exercise that many States have participated in [163, 164].

Quantitative data obtained during a nuclear forensic examination are typically associated with statistical measures of uncertainty [165]. Uncertainty, sometimes incorrectly referred to as 'error', is used to quantitatively describe the precision of measurements. Analytical techniques with higher precision produce data with smaller uncertainty, whereas lower precision techniques produce data with larger uncertainty. While the topic of statistical uncertainty in measurement is vast [166], and although nuclear forensic analysts seek to produce data with the minimal possible uncertainty [167], in many cases it is only necessary for the laboratory to produce data that meet the needs and requirements of the investigative authority. For example, if the possession of enriched uranium outside of regulatory control is in violation of a State's laws, and a low enriched uranium fuel pellet is the subject of nuclear forensic examination, it may only be necessary for the laboratory to produce data that demonstrate that the sample is enriched, taking into consideration the uncertainty of the measurement. If the sample is determined to have an enrichment of 3% 235 U, with a relatively large uncertainty of ±1.5% ²³⁵U, this measurement will have sufficiently demonstrated that the material exceeds the isotopic composition of natural uranium (0.725% ²³⁵U) if measurement accuracy can also be established through the use of standard reference materials. In this case, a more precise measurement might not be necessary to meet the needs and requirements of the investigative authority. Therefore, even relatively low-precision measurements may be sufficiently precise to address the needs and requirements of the investigative authority, and measurement accuracy can be demonstrated using standard reference materials, standard operating procedures and through other means.

When reporting measurement uncertainty, the nuclear forensic laboratory may provide sufficient information to ensure that uncertainty calculations can be reproduced if necessary. By reporting the mathematical approach used to quantify uncertainty, the investigative authority can be assured as to the degree of confidence in each datum. Reference [168] describes an approach taken by some nuclear forensic laboratories for uncertainty quantification in nuclear forensics.

8. REPORTING THE RESULTS OF A NUCLEAR FORENSIC EXAMINATION

A nuclear forensic examination typically results in the production of a series of reports containing key data on the physical, chemical, elemental and/or isotopic properties of the nuclear or radioactive material and/or associated non-nuclear material, as well as other findings

and conclusions resulting from the examination. These reported findings can then be used by the investigative authority to aid the investigation and may even be used in criminal or other legal proceedings. Reports can be written in such a way as to ensure that findings are understood by a non-technical audience, such as the investigative authority as well as others (e.g. policymakers, government officials and those involved in legal proceedings) who may consult the reports. Most laboratories prepare reports of findings from nuclear forensic examinations that are consistent with the requirements of the State's legal statutes for use during a criminal investigation.

In addition to relevant nuclear forensic data (including the respective uncertainties), as well as key conclusions, reports may contain descriptions of methods and techniques specific to the production of the data. Methods applied to a nuclear forensic examination often follow standard operating procedures as far as possible. If deviation from a standard operating procedure is necessary (e.g. the analysis of a sample of an unusual matrix), the report may state the nature of the deviations and why they were made. Standard reference material data are typically reported alongside sample data to demonstrate measurement accuracy.

In cases where more than one technique is used to analyse a particular nuclear forensic signature (e.g. uranium isotopic composition, which can be measured by gamma spectrometry, alpha spectrometry or mass spectrometry), the report may stipulate which set of instrumental data is the most reliable. The report may evaluate the consistency of observations derived from the application of multiple, independent methods. In cases where two techniques produce data for a particular nuclear forensic signature that do not agree (i.e. do not overlap within the limits of uncertainty), it is good practice for the laboratory to explain why this might have occurred and to assess the reliability of these data. Very often, it is because the first analytical technique selected was not well suited to the nuclear forensic sample matrix, and another technique was later chosen that was more appropriate.

For techniques used to measure physical nuclear forensic signatures, such as SEM, qualitative observations (e.g. descriptions of the physical appearance of a sample, or of various features observed, including scratches and other signs of damage) are commonly treated as data. It is often difficult to ascribe uncertainty to these observations, and physical descriptions may be subjective, depending on the perspective and expertise of the analyst. Maintaining objectivity is good practice for ensuring that descriptions are understood by the investigative authority and by others not involved in the nuclear forensic examination. Some States have developed a standard vocabulary for the description of nuclear or other radioactive material so that specific descriptions have well defined meanings [36]. Physical descriptions are typically paired with photographs and/or photomicrographs in order to enable the investigative authority to interpret these features in the broader context of the investigation.

In some cases, the investigative authority may request additional laboratory assistance in the interpretation of nuclear forensic data. Nuclear forensic interpretation typically involves the comparison and association of nuclear forensic signatures with information pertaining to types of material and origin and method of production of nuclear or other radioactive material, or with previous cases involving similar material. This information can be obtained from sources including: (i) existing information within the State (e.g. a national nuclear forensic library [169–171], a sealed source registry and/or a nuclear material accountancy and control system); (ii) by performing comparative analysis with similar samples available in the laboratory; (iii) through subject matter expertise.

If no additional analyses are requested by the investigative authority, samples and data generated as a result of the examination may be preserved and stored, consistent with the requirements of national laws and regulations, until the investigation has concluded.

9. INTERNATIONAL COOPERATION AND ASSISTANCE

Not all States have developed sufficient national capacity in nuclear forensics to analyse the types of nuclear or other radioactive material that might be encountered out of regulatory control. Therefore, international cooperation is a useful approach for the development of nuclear forensics in any State. States with experience in the development and implementation of a nuclear forensic capability may be willing to share their knowledge with other States, with the goal of strengthening nuclear security worldwide.

States can assist other States through the sharing of best practices in nuclear forensic analysis, through the implementation of training in nuclear forensics and through participation in bilateral or multilateral exercises involving the analysis of nuclear or other radioactive material, as well as by many other means. In addition, States with developed nuclear forensic capabilities may extend offers of assistance to other States in the examination of nuclear or other radioactive material encountered out of regulatory control, and States may accept offers of assistance. This approach is useful if the analytical needs of a nuclear forensic examination exceed the ability of a State to perform the necessary analyses or for other reasons.

The IAEA has developed a capacity building programme in nuclear forensics, designed to assist States in developing nuclear forensic capabilities. States interested in developing nuclear forensics may request expert missions. Upon such requests, the IAEA assists in the coordination of relevant national stakeholders and in the identification of national resources to be used in nuclear forensics. States may also choose to participate in the IAEA training programme, which utilizes the subject matter expertise of nuclear forensic experts. The IAEA also holds technical meetings on a periodic basis, focused on developments in nuclear forensics [172, 173]. These meetings function as forums for Member States to present developments in their nuclear forensic capabilities as well as advancements in nuclear forensic science. Member States interested in making contributions to the development of nuclear forensic science may apply for participation in IAEA coordinated research activities, which support Member States in research and development activities in nuclear forensics. The IAEA has developed several coordinated research projects in the area of nuclear forensics that have resulted in the publication of technical documents reporting the results of these projects [174, 175].

The Nuclear Forensics International Technical Working Group is another organization that administers several activities designed to assist States in developing and exercising a nuclear forensic capability. The 'Collaborative Material Exercise' series of exercises involves the analysis of nuclear or other radioactive material, enabling States to apply their nuclear forensic capability through actual examination of nuclear or other radioactive material [163]. The 'Galaxy Serpent' series of exercises consists of virtual, web-based exercises designed to illustrate the utility of national nuclear forensic libraries in addressing the needs and requirements of investigative authorities as part of investigations involving nuclear or other radioactive material out of regulatory control [176].
REFERENCES

- [1] INTERNATIONAL ATOMIC ENERGY AGENCY, Nuclear Forensics in Support of Investigations, IAEA Nuclear Security Series No. 2-G (Rev. 1), IAEA, Vienna (2015).
- [2] INTERNATIONAL ATOMIC ENERGY AGENCY, Radiological Crime Scene Management, IAEA Nuclear Security Series No. 22-G, IAEA, Vienna (2014).
- [3] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on In-Field Applications of High-Resolution Gamma Spectrometry for Analysis of Special Nuclear Material (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-GSIF.pdf.
- [4] Amendment to the Convention on the Physical Protection of Nuclear Material, INFCIRC/274/Rev. 1/Mod. 1 (Corrected), IAEA, Vienna (2021).
- [5] INTERNATIONAL ATOMIC ENERGY AGENCY, Categorization of Radioactive Sources, IAEA Safety Standards Series No. RS-G-1.9, IAEA, Vienna (2005).
- [6] SMITH, D.K., KRISTO, M.J., NIEMEYER, S., DUDDER, G.B., Documentation of a model action plan to deter illicit nuclear trafficking, J. Radioanal. Nucl. Chem. 276 (2008) 415–419.
- [7] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Laboratory Applications of High-Resolution Gamma Spectrometry (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-GSLA.pdf.
- [8] POUDEL, D., AVTANDILASHVILI, M., BERTELLI, L., KLUMPP, J.A., TOMACHEV, S.Y., Long-term retention of plutonium in the respiratory tracts of two acutely-exposed workers: estimation of bound fraction, Health Phys. 120 (2021) 258– 270.
- [9] JOVANOVIC, S.V., et al., Uncovering uranium isotopic heterogeneity of fuel pellets from the fifth collaborative materials exercise of the Nuclear Forensics International Technical Working Group, J. Radioanal. Nucl. Chem. **326** (2020) 1853–1866.
- [10] OLSEN, A.M., et al., Quantifying morphological features of α-U₃O₈ with image analysis for nuclear forensics, Anal. Chem. **89** (2017) 3177–3183.
- [11] KRAIEM, M., et al., Characterizing uranium oxide reference particles for isotopic abundances and uranium mass by single particle isotope dilution mass spectrometry, Anal. Chim. Acta 748 (2012) 37–44.
- [12] VARGA, Z., WALLENIUS, M., NICHOLL, A., MAYER, K., Assessment of uranium inhomogeneity and isotope imaging for nuclear forensics, Spectrochim. Acta B 171 (2020) 105920.

- [13] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on High-Resolution Gamma Spectrometry (2016), https://www.nfitwg.org/pdfs/ITWG-INFL-GSOV.pdf.
- [14] LAKOSI, L., et al., Gamma spectrometry in the ITWG CMX-4 exercise, J. Radioanal. Nucl. Chem. 315 (2018) 409–416.
- [15] APOSTOL, A.I., et al., Characterization of californium sources by gamma spectrometry: relevance for nuclear forensics, J. Radioanal. Nucl. Chem. **321** (2019) 405–412.
- [16] RAMEBÄCK, H., et al., Basic characterization of ²³³U: determination of age and ²³²U content using sector field ICP-MS, gamma spectrometry and alpha spectrometry, Nucl. Instrum. Methods Phys. Res. Sect. B 266 (2008) 807–812.
- [17] ZSIGRAI, J., NGUYEN, T.C., BERLIZOV, A., Gamma-spectrometric determination of ²³²U in uranium-bearing materials, Nucl. Instrum. Methods Phys. Res. Sect. B **359** (2015) 137–144.
- [18] CHOI, H-D., KIM, J., Basic characterization of uranium by high-resolution gamma spectroscopy, Nucl. Eng. Technol. 50 (2018) 929–936.
- [19] TANDON, L., et al., Establishing reactor operations from uranium targets used for the production of plutonium, J. Radioanal. Nucl. Chem. 282 (2009) 573–579.
- [20] CLOUVAS, A., XANTHOS, S., TAKOUDIS, G., POTIRIADIS, C., SILVA, J., In situ gamma spectrometry measurements and Monte Carlo computations for the detection of radioactive sources in scrap metal, Health Phys. 88 (2005) 154–162.
- [21] ELISH, E., et al., Characterization of an orphan Ra-Be neutron source, J. Radioanal. Nucl. Chem. 329 (2021) 51–60.
- [22] NGUYEN, C.T., Age-dating of highly enriched uranium by γ-spectrometry, Nucl. Instrum. Methods Phys. Res. Sect. B 229 (2005) 103–110.
- [23] NGUYEN, C.T., ZSIGRAI, J., Basic characterization of highly enriched uranium by gamma spectrometry, Nucl. Instrum. Methods Phys. Res. Sect. B **246** (2006) 417–424.
- [24] WALLENIUS, M., MAYER, K., Age determination of plutonium material in nuclear forensics by thermal ionisation mass spectrometry, Fresenius J. Anal. Chem. 366 (2000) 234–238.
- [25] WALLENIUS, M., MAYER, K., RAY, I., Nuclear forensic investigations: two case studies, Forensic Sci. Int. 156 (2006) 55–62.
- [26] LÜTZENKIRCHEN, K., et al., Nuclear forensics on uranium fuel pellets, Radiochim. Acta 107 (2019) 635–643.

- [27] CHAMBERLAIN, D., et al., "Resources and Forensics Signatures to Help Determine the Origin of Sealed Radiological Sources", International Conference on Advances in Nuclear Forensics: Countering the Evolving Threat of Nuclear and Other Radioactive Material out of Regulatory Control, Vienna, 7–10 July 2014 (Proc. Int. Conf. Vienna, 2014), IAEA, Vienna (2014), https://www-pub.iaea.org/MTCD/Publications/PDF/Supplementary Materials/P1706/Technical Session 3C.pdf.
- [28] GRIFFITHS, G., et al., Physical characterization of uranium oxide pellets and powder applied in the Nuclear Forensics International Technical Working Group Collaborative Materials Exercise 4, J. Radioanal. Nucl. Chem. **315** (2018) 365–378.
- [29] THOMPSON, N.B.A., et al., Objective colour analysis from digital images as a nuclear forensic tool, Forensic Sci. Int. 319 (2021) 110678.
- [30] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Characteristic Parameters of Uranium Dioxide (UO₂) Fuel Pellets (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-UOFP.pdf.
- [31] MAYER, K., WALLENIUS, M., RAY, I., Nuclear forensics—A methodology providing clues on the origin of illicitly trafficked nuclear materials, Analyst **130** (2005) 433–441.
- [32] SHABAN, S.E., EL-TAYEBANY, R.A., HAZZAA, M.H., Construction of a developed software for nuclear forensics of micro particles, J. Radiat. Res. Appl. Sci. 12 (2019) 101–105.
- [33] LY, C., VACHET, et al., Determining uranium ore concentrates and their calcination products via image classification of multiple magnifications, J. Nucl. Mater. 533 (2020) 152082.
- [34] SCHWERDT, I.J., et al., Nuclear forensics investigation of morphological signatures in the thermal decomposition of uranyl peroxide, Talanta **176** (2018) 284-292.
- [35] FONGARO, L., HO, D.M.L, KVAAL, K., MAYER, K., RONDINELLA, V.V., Application of the angle measure technique as image texture analysis method for the identification of uranium ore concentrate samples: new perspective in nuclear forensics, Talanta 152 (2016) 463–474.
- [36] TAMASI, A.L., et al., A lexicon for consistent description of material images for nuclear forensics, J. Radioanal. Nucl. Chem. 307 (2016) 1611–1619.
- [37] STEEB, J.L., et al., Impact of an external radiation field on handheld XRF measurements for nuclear forensics applications, J. Radioanal. Nucl. Chem. 307 (2016) 751–760.
- [38] GRANT, P.M., et al., Nuclear forensics in law enforcement applications, J. Radioanal. Nucl. Chem. 235 (1998) 129–132.

- [39] KEATLEY, A.C., SCOTT, T.B., DAVIS, S., JONES, C.P., TURNER, P., An investigation into heterogeneity in a single vein-type uranium ore deposit: implications for nuclear forensics, J. Environ. Radioact. 150 (2015) 75–85.
- [40] FAHEY, A.J., ZEISSLER, C.J., NEWBURY, D.E., DAVIS, J., LINDSTROM, R.M., Postdetonation nuclear debris for attribution, Proc. Natl. Acad. Sci. U.S.A. 107 (2010) 20207–20212.
- [41] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Powder X-ray Diffraction (XRD)—General Overview (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-PXRD.pdf.
- [42] ELOIRDI, R., HO, D.M.L., MAYER, K., CACIUFFO, R., FANGHÄNEL, T., Investigation of ammonium diuranate calcination with high-temperature X-ray diffraction, J. Mater. Sci. 49 (2014) 8436–8443.
- [43] TRACY, C.L., CHEN, C-H., PARK, S., DAVISSON, M.L., EWING, R.C., Measurement of UO₂ surface oxidation using grazing-incidence x-ray diffraction: implications for nuclear forensics, J. Nucl. Mater. 502 (2018) 68–75.
- [44] KRISTO, M.J., TUMEY, S.J., The state of nuclear forensics, Nucl. Instrum. Methods Phys. Res. Sect. B 294 (2013) 656–661.
- [45] KUCHKIN, A., et al., Contribution of bulk mass spectrometry isotopic analysis to characterization of materials in the framework of CMX-4, J. Radioanal. Nucl. Chem. 315 (2018) 435–441.
- [46] KRACHLER, M., ALVAREZ-SARANDES, R., VAN WINCKEL, S., Challenges in the quality assurance of elemental and isotopic analyses in the nuclear domain benefitting from high resolution ICP-OES and sector field ICP-MS, J. Radioanal. Nucl. Chem. 304 (2015) 1201–1209.
- [47] PANDEY, A., et al., Analysis of Th and U in thorium-based mixed-oxide fuel using wavelength dispersive X-ray fluorescence spectrometer, J. Radioanal. Nucl. Chem. 319 (2019) 775–781.
- [48] ORIHASHI, Y., HIRATA, T., Rapid quantitative analysis of Y and REE abundances in XRF glass bead for selected GSJ reference rock standards using Nd-YAG 266 nm UV laser ablation ICP-MS, Geochem. J. 37 (2003) 401–412.
- [49] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Elemental Assay—U Titration (2017), https://www.nfitwg.org/pdfs/ITWG-Guideline-U-Titration.pdf.

- [50] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Elemental Assay—Pu Titration (2017), https://www.nfitwg.org/pdfs/ITWG-Guideline-Pu-Titration.pdf.
- [51] DAVIES, W., GRAY, W., A rapid and specific titrimetric method for the precise determination of uranium using iron(II) sulphate as reductant, Talanta 11 (1964) 1203– 1211.
- [52] NELWAMONDO, A.N., et al., Uranium assay and trace element analysis of the fourth collaborative material exercise samples by the modified Davies-Gray method and the ICP-MS/OES techniques, J. Radioanal. Nucl. Chem. **315** (2018) 379–394.
- [53] MATHEW, K., et al., Uranium assay determination using Davies and Gray titration: an overview and implementation of GUM for uncertainty evaluation, J. Radioanal. Nucl. Chem. 282 (2009) 939–944.
- [54] RONDAHL, S.H., et al., Comparing results of X-ray diffraction, μ-Raman spectroscopy and neutron diffraction when identifying chemical phases in seized nuclear material, during a comparative nuclear forensics exercise, J. Radioanal. Nucl. Chem. **315** (2018) 395–408.
- [55] MAYER, K., WALLENIUS, M., VARGA, Z., Interviewing a silent (radioactive) witness through nuclear forensic analysis, Anal. Chem. 87 (2015) 11605–11610.
- [56] GILLENS, A.R., POWELL, B.A., Rapid quantification of TBP and TBP degradation product ratios by FTIR-ATR, J. Radioanal. Nucl. Chem. 296 (2013) 859–868.
- [57] BÜRGER, S., et al., Quantifying multiple trace elements in uranium ore concentrates: an interlaboratory comparison, J. Radioanal. Nucl. Chem. **301** (2014) 711–729.
- [58] VARGA, Z., WALLENIUS, M., MAYER, K., Origin assessment of uranium ore concentrates based on their rare-earth elemental impurity pattern, Radiochim. Acta 98 (2010) 771–778.
- [59] VARGA, Z., KATONA, R., STEFÁNKA, Z., WALLENIUS, M., MAYER, K., NICHOLL, A., Determination of rare-earth elements in uranium-bearing materials by inductively coupled plasma mass spectrometry, Talanta 80 (2010) 1744–1749.
- [60] MERCADIER, J., et al., Origin of uranium deposits revealed by their rare earth element signature, Terra Nova 23 (2011) 264–269.
- [61] KEEGAN, E., et al., Nuclear forensic analysis of an unknown uranium ore concentrate sample seized in a criminal investigation in Australia, Forensic Sci. Int. 240 (2014) 111– 121.

- [62] FRIMMEL, H.E., SCHEDEL, S., BRÄTZ, H., Uraninite chemistry as forensic tool for provenance analysis, App. Geochem. 48 (2014) 104–121.
- [63] SPANO, T.L., SIMONETTI, A., BALBONI, E., DORAIS, C., BURNS, P.C., Trace element and U isotope analysis of uraninite and ore concentrate: applications for nuclear forensic investigations, App. Geochem. 84 (2017) 277–285.
- [64] SERBAN, A., et al., Validation of in-house procedure for monitoring rare earth elements and lead elemental impurities in uranium materials, J. Radioanal. Nucl. Chem. 330 (2021) 91–101.
- [65] VARGA, Z., WALLENIUS, M., MAYER, K., MEPPEN, M., Analysis of uranium ore concentrates for origin assessment, Proc. Radiochim. Acta 1 (2011) 1–4.
- [66] MADZUNYA, D., UUSHONA, V., MATHUTHU, M., HEIKE, W., Rare earth elements in uranium ore deposits from Namibia: a nuclear forensics tool, J. Environ. Radioact. 237 (2021) 106668.
- [67] MAYER, K., WALLENIUS, M., VARGA, Z., Nuclear forensic science: correlating measurable material parameters to the history of nuclear material, Chem. Rev. 113 (2013) 884–900.
- [68] BADAUT, V., WALLENIUS, M., MAYER, K., Anion analysis in uranium ore concentrates by ion chromatography, J. Radioanal. Nucl. Chem. 280 (2009) 57–61.
- [69] KUMAR, J.R., KIM, J-S., LEE, J-Y., YOON, H-S., A brief review on solvent extraction of uranium from acidic solutions, Sep. Purif. Rev. 40 (2011) 77–125.
- [70] KEEGAN, E., WALLENIUS, M., MAYER, K., VARGA, Z., RASMUSSEN, G., Attribution of uranium ore concentrates using elemental and anionic data, Appl. Geochem. 27 (2012) 1600–1609.
- [71] KLOOSTERMAN, J.L., Application of boron and gadolinium burnable poison particles in UO₂ and PuO₂ fuels in HTRs, Ann. Nucl. Energy **30** (2003) 1807–1819.
- [72] BARANOV, V.G., et al., Sintering parameters for uranium-gadolinium oxide fuel pellets, Atom. Energy 112 (2012) 303–306.
- [73] COWAN, G.A., ADLER, H.H., The variability of the natural abundance of ²³⁵U, Geochim. Cosmochim. Acta 40 (1976) 1487–1490.
- [74] GOLDMANN, A., BRENNECKA, G., NOORDMANN, J., WEYER, S., WADHWA, M., The uranium isotopic composition of the Earth and the Solar System, Geochim. Cosmochim. Acta 148 (2015) 145–158.

- [75] KU, T-L., MATHIEU, G.G., KNAUSS, K.G., Uranium in open ocean: concentration and isotopic composition, Deep-Sea Res. 24 (1977) 1005–1017.
- [76] MIRONOV, V.P., MATUSEVICH, J.L., KUDRJASHOV, V.P., BOULYGA, S.F., BECKER, J.S., Determination of irradiated reactor uranium in soil samples in Belarus using ²³⁶U as irradiated uranium tracer, J. Environ. Monit. 4 (2002) 997–1002.
- [77] MURAMATSU, Y., UCHIDA, S., TAGAMI, K., YOSHIDA, S., FUJIKAWA, T., Determination of plutonium concentration and its isotopic ratio in environmental materials by ICP-MS after separation using ion-exchange and extraction chromatography, J. Anal. At. Spectrom. 14 (1999) 859–865.
- [78] POINTURIER, F., HUBERT, A., BAGLAN, N., HEMET, P., Evaluation of a new generation quadrupole-based ICP-MS for uranium isotopic measurements in environmental samples, J. Radioanal. Nucl. Chem. 276 (2008) 505–511.
- [79] VARGA, Z., SURANYI, G., Detection of previous neutron irradiation and reprocessing of uranium materials for nuclear forensic purposes, Appl. Radiat. Isot. 67 (2009) 516– 522.
- [80] POINTURIER, F., HUBERT, A., FAURE, A-L., HEMET, P., POTTIN, A-C., Polyatomic interferences in plutonium determination in the femtogram range by doublefocusing sector-field ICP-MS, J. Anal. At. Spectrom. 26 (2011) 1474–1480.
- [81] SHI, Y., COLLINS, R., BROOME, C., Determination of uranium, thorium and plutonium isotopes by ICP-MS, J. Radioanal. Nucl. Chem. 296 (2013) 509–515.
- [82] BOULYGA, S.F., KOEPF, A., KONEGGER-KAPPEL, S., MÁCSIK, Z., STADELMANN, G., Uranium isotope analysis by MC-ICP-MS in sub-ng sized samples, J. Anal. At. Spectrom. **31** (2016) 2272–2284.
- [83] EPPICH, G.R., et al., Plutonium assay and isotopic composition measurements in nuclear safeguards samples by inductively coupled plasma mass spectrometry, J. Anal. At. Spectrom. 34 (2019) 1154–1165.
- [84] RICHTER, S., ALONSO, A., WELLUM., R., TAYLOR, P.D.P., The isotopic composition of commercially available uranium chemical reagents, J. Anal. At. Spectrom. 14 (1999) 889–891.
- [85] RICHTER, S., GOLDBERG, S.A., Improved techniques for high accuracy isotope ratio measurements of nuclear materials using thermal ionization mass spectrometry, Int. J. Mass Spectrom. 229 (2003) 181–197.
- [86] PEREIRA DE OLIVEIRA, O., et al., Evaluation of the $n(^{235}\text{U})/n(^{238}\text{U})$ isotope ratio measurements in a set of uranium samples by thermal ionization mass spectrometry, Int. J. Mass Spectrom. **246** (2005) 35–42.

- [87] RICHTER, S., et al., Evaluating the status of uranium isotope ratio measurements using an inter-laboratory comparison campaign, Int. J. Mass Spectrom. **264** (2007) 184–190.
- [88] RICHTER, S., et al., Improvements in routine uranium isotope ratio measurements using the modified total evaporation method for multi-collector thermal ionization mass spectrometry, J. Anal. At. Spectrom. 26 (2011) 550–564.
- [89] STURM, M., RICHTER, S., AREGBE, Y., WELLUM, R., PROHASKA, T., Optimized chemical separation and measurement by TE TIMS using carburized filaments for uranium isotope ratio measurements applied to plutonium chronometry, Anal. Chem. 88 (2016) 6223–6230.
- [90] KRISTO, M.J., et al., The application of radiochronometry during the 4th collaborative materials exercise of the nuclear forensics international technical working group (ITWG), J. Radioanal. Nucl. Chem. **315** (2018) 425–434.
- [91] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Alpha Spectrometry (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-ALPH.pdf.
- [92] SCHULZ, R.K., WINK, G.T., FUJII, L.M., Sequential determination of neptunium-237, plutonium-238, plutonium-239, 240, americium-241, and curium-244 in plant material, Soil Sci. 132 (1981) 71–76.
- [93] PEŃKIN, M., et al., Isotopic composition of commercially available uranium chemicals and elemental analysis standards, J. Radioanal. Nucl. Chem. **316** (2018) 791–798.
- [94] DE LA GARZA, A., GARRETT, G.A., MURPHY, J.E., Multicomponent isotope separation in cascades, Chem. Eng. Sci. **15** (1961) 188–209.
- [95] HANSON, S.K., POLLINGTON, A.D., Measuring signatures of fuel irradiation in large particle samples, J. Anal. At. Spectrom. 36 (2021) 1018–1027.
- [96] WALLENIUS, M., et al., Nuclear forensic investigations with a focus on plutonium, J. Alloy. Compd. 444-445 (2007) 57–62.
- [97] MAYER, K., WALLENIUS, M., HEDBERG, M., LÜTZENKIRCHEN, K., Unveiling the history of seized plutonium through nuclear forensic investigations, Radiochim. Acta 97 (2009) 261–264.
- [98] ROBEL, M., et al., A composite position independent monitor of reactor fuel irradiation using Pu, Cs, and Ba isotope ratios, J. Environ. Radioactiv. 195 (2018) 9–19.
- [99] OSBORN, J.M., KITCHER, E.D., BURNS, J.D., FOLDEN III, C.M., CHIRAYATH, S.S., Nuclear forensics methodology for reactor-type attribution of chemically separated plutonium, Nucl. Technol. 201 (2018) 1–10.

- [100] NICOLAOU, G., BIEGALSKI, S.R., Discrimination of plutonium from thermal reactors in the frame of nuclear forensics, J. Radioanal. Nucl. Chem. **317** (2018) 559–564.
- [101] RIM, J.H., et al., Determination of origin and intended use of plutonium metal using nuclear forensic techniques, Forensic Sci. Int. **273** (2017) e1–e9.
- [102] KITCHER, E.D., OSBORN, J.M., CHIRAYATH, S.S., Sensitivity studies on a novel nuclear forensics methodology for source reactor-type discrimination of separated weapons grade plutonium, Nucl. Eng. Technol. 51 (2019) 1355–1364.
- [103] OSBORN, J.M., et al., Experimental validation of a nuclear forensics methodology for source reactor-type discrimination of chemically separated plutonium, Nucl. Eng. Technol. 51 (2019) 384–393.
- [104] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline on Age Dating (Production Date Determination) (2016), https://www.nf-itwg.org/pdfs/ITWG-INFL-ADPD.pdf.
- [105] LAMONT, S.P., HALL, G., Uranium age determination by measuring the ²³⁰Th/²³⁴U ratio, J. Radioanal. Nucl. Chem. 264 (2005) 423–427.
- [106] VARGA, Z., et al., Validation of reference materials for uranium radiochronometry in the frame of nuclear forensic investigations, Appl. Radiat. Isot. **102** (2015) 81–86.
- [107] VARGA, Z., SURANYI, G., Production date determination of uranium-oxide materials by inductively coupled plasma mass spectrometry, Anal. Chim. Acta 599 (2007) 16–23.
- [108] WILLIAMS, R.W., GAFFNEY, A.M., ²³⁰Th-²³⁴U model ages of some uranium standard reference materials, Proc. Radiochim. Acta 1 (2011) 31–35.
- [109] GAFFNEY, A.M., et al., Round-robin ²³⁰Th-²³⁴U age dating of bulk uranium for nuclear forensics, J. Radioanal. Nucl. Chem. **307** (2016) 2055–2060.
- [110] OKUBO, A., SHINOHARA, N., MAGARA, M., Uranium age-dating using in-situ isotope ratios by thermal ionization mass spectrometry for nuclear forensics, J. Radioanal. Nucl. Chem. **314** (2017) 231–234.
- [111] MORGENSTERN, A., APOSTOLIDIS, C., MAYER, K., Age determination of highly enriched uranium: separation and analysis of ²³¹Pa, Anal. Chem. 74 (2002) 5513–5516.
- [112] EPPICH, G.R., WILLIAMS, R.W., GAFFNEY, A.M., SCHORZMAN, K.C., ²³⁵U-²³¹Pa age dating of uranium materials for nuclear forensic investigations, J. Anal. At. Spectrom. 28 (2013) 666-674.

- [113] TREINEN, K.C., et al., Improved protactinium spike calibration method applied to ²³¹Pa-²³⁵U age-dating of certified reference materials for nuclear forensics, J. Radioanal. Nucl. Chem. **318** (2018) 209–219.
- [114] TREINEN, K.C., et al., Evaluating uranium radiochronometry by single-collector mass spectrometry for nuclear forensics: a multi-instrument investigation, J. Radioanal. Nucl. Chem. 322 (2019) 1627–1640.
- [115] HIGGINSON, M., et al., Development of rapid methodologies for uranium age dating, J. Radioanal. Nucl. Chem. **318** (2018) 157–164.
- [116] CHEN, Y., et al., Age determination for uranium standard samples by ²³¹Pa/²³⁵U radiochronometer, J. Radioanal. Nucl. Chem. **322** (2019) 1605–1609.
- [117] DENTON, J.S., et al., International cooperation in age-dating uranium standards for nuclear forensics using the ²³¹Pa/²³⁵U radiochronometer, J. Radioanal. Nucl. Chem. **324** (2020) 705–714.
- [118] CHEN, Y., et al., Studies on the age determination of trace plutonium, J. Radioanal. Nucl. Chem. 281 (2009) 675–678.
- [119] STURM, M., et al., Evaluation of chronometrics in plutonium age determination for nuclear forensics: What if the 'Pu/U clocks' do not match?, J. Radioanal. Nucl. Chem. 302 (2014) 399–411.
- [120] GAFFNEY, A.M., et al., A case study in plutonium radiochronometry using multiple isotope systems, J. Radioanal. Nucl. Chem. 318 (2018) 287–295.
- [121] VARGA, Z., NICHOLL, A., WALLENIUS, M., MAYER, K., Plutonium age dating (production date measurement) by inductively coupled plasma mass spectrometry, J. Radioanal. Nucl. Chem. **307** (2016) 1919–1926.
- [122] KAYZAR, T.M., WILLIAMS, R.W., Developing ²²⁶Ra and ²²⁷Ac age-dating techniques for nuclear forensics to gain insight from concordant and non-concordant radiochronometers, J. Radioanal. Nucl. Chem. **307** (2016) 2061–2068.
- [123] ROLISON, J.M., TREINEN, K.C., MCHUGH, K.C., GAFFNEY, A.M., WILLIAMS, R.W., Application of the ²²⁶Ra-²³⁰Th-²³⁴U and ²²⁷Ac-²³¹Pa-²³⁵U radiochronometers to uranium certified reference materials, J. Radioanal. Nucl. Chem. **314** (2017) 2459–2467.
- [124] CHENG, H., EDWARDS, R.L., MURRELL, M.T., BENJAMIN, T.M., Uraniumthorium-protactinium dating systematics, Geochim. Cosmochim. Acta 62 (1998) 3437– 3452.

- [125] MATHEW, K., et al., Intercomparison of the radio-chronometric ages of plutoniumcertified reference materials with distinct isotopic compositions, Anal. Chem. 91 (2019) 11643–11652.
- [126] VARGA, Z., WALLENIUS, M., MAYER, K., KEEGAN, E., MILLET, S., Application of lead and strontium isotope ratio measurements for the origin assessment of uranium ore concentrates, Anal. Chem. 81 (2009) 8327–8334.
- [127] KEEGAN, E., et al., The provenance of Australian uranium ore concentrates by elemental and isotopic analysis, App. Geochem. 23 (2008) 765–777.
- [128] KRAJKO, J., VARGA, Z., YALCINTAS, E., WALLENIUS, M., MAYER, K., Application of neodymium isotope ratio measurements for the origin assessment of uranium ore concentrates, Talanta 129 (2014) 499–504.
- [129] MIGEON, V., BOURDON, B., PILI, E., FITOUSSI, C., An enhanced method for molybdenum separation and isotopic determination in uranium-rich materials and geological samples, J. Anal. At. Spectrom. **30** (2015) 1988–1996.
- [130] ROLISON, J.M., et al., Molybdenum isotope compositions of uranium ore concentrates by double spike MC-ICP-MS, App. Geochem. 103 (2019) 97–105.
- [131] SHOLLENBERGER, Q.R., BORG, L.E., RAMON, E.C., SHARP, M.A., BRENNECKA, G.A., Samarium isotope compositions of uranium ore concentrates: a novel nuclear forensic signature, Talanta 221 (2021) 121431.
- [132] KAYZAR-BOGGS, T.M., et al., Exploring the use of thorium isotope compositions and concentrations as nuclear forensic signatures for uranium ore concentrates, J. Radioanal. Nucl. Chem. **327** (2021) 877–889.
- [133] ISNARD, H., et al., Determination of ⁹⁰Sr/²³⁸U ratio by double isotope dilution inductively coupled plasma mass spectrometer with multiple collection in spent nuclear fuel samples with in situ ⁹⁰Sr/⁹⁰Zr separation in a collision-reaction cell, Spectrochim. Acta B **61** (2006) 150–156.
- [134] NAGY, P., VAJDA, N., SZIKLAI-LASZLO, I., KOVACS-SZELES, E., SIMONITS, A., Determination of ¹³⁵Cs in nuclear power plant wastes by ICP-MS and k₀-NAA, J. Radioanal. Nucl. Chem. **300** (2014) 615–627.
- [135] OSVATH, S., et al., Determination of ⁹³Zr in nuclear power plant wastes, J. Radioanal. Nucl. Chem. **314** (2017) 31–38.
- [136] GANEA, L.M., APOSTOL, A.I., PANTELICA, A., Non-destructive analysis of purebeta emitters: applications in nuclear forensics, J. Radioanal. Nucl. Chem. 322 (2019) 1641–1644.

- [137] GLENNON, K.J., et al., Measuring key Sm isotope ratios in irradiated UO₂ for use in plutonium discrimination nuclear forensics, J. Radioanal. Nucl. Chem. **320** (2019) 405– 414.
- [138] GLENNON, K.J., et al., Nuclear forensics methodology identifies legacy plutonium from the Manhattan Project, J. Radioanal. Nucl. Chem. **330** (2021) 57–65.
- [139] MIGEON, V., FITOUSSI, C., PILI, E., BOURDON, B., Molybdenum isotope fractionation in uranium oxides and during key processes of the nuclear fuel cycle: towards a new nuclear forensic tool, Geochim. Cosmochim. Acta 279 (2020) 238–257.
- [140] PAJO, L., MAYER, K., KOCH, L., Investigation of the oxygen isotopic composition in oxidic uranium compounds as a new property in nuclear forensic science, Fresenius J. Anal. Chem. 371 (2001) 348–352.
- [141] OERTER, E.J., et al., Hydrogen and oxygen stable isotope composition of water in metaschoepite mineralization on U₃O₈, App. Geochem. **112** (2020) 104469.
- [142] OERTER, E.J., et al., Stable isotope signatures of hydration water in secondary mineralization on UO₂, Talanta **226** (2021) 122096.
- [143] KLOSTERMAN, M.R., OERTER, E.J., CHAKRABORTY, S., SINGLETON, M.J., MCDONALD IV, L.W., Fractionation of oxygen isotopes in uranium oxides during peroxide precipitation and dry air calcination, ACS Earth Space Chem. 5 (2021) 1622– 1630.
- [144] KLOSTERMAN, M.R., et al., Oxygen kinetic isotope effects in the thermal decomposition and reduction of ammonium diuranate, ACS Omega 6 (2021) 30856– 30864.
- [145] STEBELKOV, V., ELANTYEV, I., HEDBERG, M., WALLENIUS, M., FAURE, A-L., Determination of isotopic composition of uranium in the CMX-4 samples by SIMS, J. Radioanal. Nucl. Chem. **315** (2018) 417–423.
- [146] KIPS, R., WEBER, P.K., KRISTO, M.J., JACOBSEN, B., RAMON, E.C., Microscale isotopic variation in uranium fuel pellets with implications for nuclear forensics, Anal. Chem. 91 (2019) 11598–11605.
- [147] FAURE, A-L., DALGER, T., Age dating of individual micrometer-sized uranium particles by secondary ion mass spectrometry: an additional fingerprint for nuclear safeguards purposes, Anal. Chem. 89 (2017) 6663–6669.
- [148] POINTURIER, F., HUBERT, A., POTTIN, A-C., Performance of laser ablation: quadrupole-based ICP-MS coupling for the analysis of single micrometric uranium particles, J. Radioanal. Nucl. Chem. 296 (2013) 609–616.

- [149] RUSH, L.A., CLIFF, J.B., REILLY, D.D., DUFFIN, A.M., MENONI, C.S., Isotopic heterogeneity imaged in a uranium fuel pellet with extreme ultraviolet laser ablation and ionization time-of-flight mass spectrometry, Anal. Chem. 93 (2021) 1016-1024.
- [150] INTERNATIONAL ORGANIZATION FOR STANDARDIZATION, General requirements for the competence of testing and calibration laboratories, ISO/IEC 17025:2017, International Organization for Standardization, Geneva (2017).
- [151] LEGGITT, J., et al., Nuclear forensics—metrological basis for legal defensibility, J. Radioanal. Nucl. Chem. 282 (2009) 997–1001.
- [152] MATHEW, K., MASON, P., VOEKS, A., NARAYANAN, U., Uranium isotope abundance ratios in natural uranium metal certified reference material 112-A, Int. J. Mass Spectrom. 315 (2012) 8–14.
- [153] INN, K.G.W., et al., The urgent requirement for new radioanalytical certified reference materials for nuclear safeguards, forensics, and consequence management, J. Radioanal. Nucl. Chem. 296 (2013) 5–22.
- [154] RICHTER, S., et al., Re-certification of a series of uranium isotope reference materials: IRMM-183, IRMM-184, IRMM-185, IRMM-186 and IRMM-187, Int. J. Mass Spectrom. 247 (2005) 37–39.
- [155] RICHTER, S., et al., A new series of uranium isotope reference materials for investigating the linearity of secondary electron multipliers in isotope mass spectrometry, Int. J. Mass Spectrom. 281 (2009) 115–125.
- [156] RICHTER, S., et al., New average values for the $n(^{238}\text{U})/n(^{235}\text{U})$ isotope ratios of natural uranium standards, Int. J. Mass Spectrom. **295** (2010) 94–97.
- [157] KRACHLER, M., VARGA, Z., NICHOLL, A., MAYER, K., Analytical considerations in the determination of uranium isotope ratios in solid uranium materials using laser ablation multi-collector ICP-MS, Anal. Chim. Acta X 2 (2019) 100018.
- [158] FITZGERALD, R., INN, K.G.W., HORGAN, C., How old is it?—²⁴¹Pu/²⁴¹Am nuclear forensic chronology reference materials, J. Radioanal. Nucl. Chem. **307** (2016) 2521– 2528.
- [159] LAVELLE, K.B., ESSEX, R.M., CARNEY, K.P., CESSNA, J.T., HEXEL, C.R., A reference material for evaluation of ¹³⁷Cs radiochronometric measurements, J. Radioanal. Nucl. Chem. **318** (2018) 195–208.
- [160] ESSEX, R.M., et al., A new highly enriched ²³³U reference material for improved simultaneous determination of uranium amount and isotope amount ratios in trace level samples, Talanta **221** (2021) 121638.

- [161] KAYZAR-BOGGS, T.M., et al., An interlaboratory collaboration to determine consensus ²³¹Pa/²³⁵U model ages of a uranium certified reference material for nuclear forensics, J. Radioanal. Nucl. Chem. **323** (2020) 1189–1195.
- [162] DORAIS, C., SIMONETTI, A., CORCORAN, L., SPANO, T.L., BURNS, P.C., Happy Jack uraninite: a new reference material for high spatial resolution analysis of U-rich materials, Geostand. Geoanal. Res. 44 (2019) 125–132.
- [163] SCHWANTES, J.M., MARSDEN, O., REILLY, D., Fourth collaborative materials exercise of the Nuclear Forensics International Technical Working Group, J. Radioanal. Nucl. Chem. 315 (2018) 347–352.
- [164] BRANDIS, M., et al., Morphological and chemical characterization of uranium and cerium nuclear forensics samples, J. Nucl. Mater. **555** (2021) 153109.
- [165] NUCLEAR FORENSICS INTERNATIONAL TECHNICAL WORKING GROUP, ITWG Guideline: Importance of Uncertainty in Nuclear Forensics Measurements (2017), https://www.nf-itwg.org/pdfs/ITWG-Guideline-Uncertainty.pdf.
- [166] POMME, S., JEROME, S.M., VENCHIARUTTI, C., Uncertainty propagation in nuclear forensics, Appl. Radiat. Isot. 89 (2014) 58–64.
- [167] MATHEW, K.J., ESSEX, R.M., GRADLE, C., NARAYANAN, U., Uncertainties achievable for uranium isotope-amount ratios: estimates based on the precision and accuracy of recent characterization measurements, J. Radioanal. Nucl. Ch. 305 (2015) 277–282.
- [168] JCMG:2008—Evaluation of Measurement Data. Guide to the Expression of Uncertainty in Measurement, Joint Committee for Guides in Metrology, Working Group 1, (2008).
- [169] INTERNATIONAL ATOMIC ENERGY AGENCY, Development of a National Nuclear Forensics Library: A System for the Identification of Nuclear or Other Radioactive Material out of Regulatory Control, IAEA-TDL-009, IAEA, Vienna (2018).
- [170] DAVYDOV, J., DION, H., LAMONT, S., HUTCHEON, I., ROBEL, M., Leveraging existing information for use in a National Nuclear Forensics Library (NNFL), J. Radioanal. Nucl. Chem. **307** (2016) 2389–2395.
- [171] KIMURA, Y., SHINOHARA, N., FUNATAKE, Y., Development of prototype nuclear forensics library for nuclear materials and radioisotopes in Japan Atomic Energy Agency, 5th International Symposium on Innovative Nuclear Energy Systems, INES-5, 31 October–2 November 2016, Tokyo Institute of Technology, Energy Procedia 131 (2017) 239–245.

- [172] INTERNATIONAL ATOMIC ENERGY AGENCY, Nuclear Forensics: Beyond the Science; Summary of a Technical Meeting, IAEA-TECDOC-1896, IAEA, Vienna (2020).
- [173] INTERNATIONAL ATOMIC ENERGY AGENCY, Advances in Nuclear Forensics: Countering the Evolving Threat of Nuclear and Other Radioactive Material out of Regulatory Control. Summary of an International Conference Organized by the International Atomic Energy Agency and Held in Vienna, 7–10 July 2014, IAEA Proceedings Series, IAEA, Vienna (2015).
- [174] INTERNATIONAL ATOMIC ENERGY AGENCY, Application of Nuclear Forensics in Combating Illicit Trafficking of Nuclear and Other Radioactive Material, IAEA-TECDOC-1730, IAEA, Vienna (2014).
- [175] INTERNATIONAL ATOMIC ENERGY AGENCY, Identification of High Confidence Nuclear Forensics Signatures: Results of a Coordinated Research Project and Related Research, IAEA-TECDOC-1820, IAEA, Vienna (2017).
- [176] BORGARDT, J.D., MARKS, N., EL-JABY, A., LAMONT, S., Results from the third Galaxy Serpent tabletop exercise demonstrating the utility of nuclear forensics libraries in support of an investigation, J. Radioanal. Nucl. Chem. 322 (2019) 1645–1656.

BIBLIOGRAPHY

BARSHICK, C.M., DUCKWORTH, D.C., SMITH, D.H. (Eds), Inorganic Mass Spectrometry: Fundamentals and Applications, CRC Press, Boca Raton (2000) 528 pp.

BECKER, J.S., Inorganic Mass Spectrometry: Principles and Applications, John Wiley and Sons, Chichester (2007) 514 pp.

BECKHOFF, B., KANNGIESSER B., LANGHOFF, N., WEDELL, R., WOLFF, H. (Eds), Handbook of Practical X-ray Fluorescence Analysis, Springer, Berlin (2006) 863 pp.

BENNINGHOVEN, A., OKANO, J., SHIMIZU, R., WERNER, H.W. (Eds), Secondary Ion Mass Spectrometry SIMS IV: Proceedings of the Fourth International Conference, Osaka, Japan, November 13–19, 1983 (Springer Series in Chemical Physics, 36), Springer, Berlin (1984) 506 pp.

BISH, D.L., POST, J.E. (Eds), Modern Powder Diffraction, Reviews in Mineralogy Vol. 20, Mineralogical Society of America, Chantilly, VA (1989) 369 pp.

BOSS, C.B., FREDEEN, K.J., Concepts, Instrumentation and Techniques in Inductively Coupled Plasma Optical Emission Spectrometry, 3rd edn, Perkin Elmer, Waltham, MA (1997) 119 pp.

CIGNONI, P., SCOPIGNO, R., Sampled 3D models for CH applications: a viable and enabling new medium or just a technological exercise?, ACM J. Comput. Cult. Herit. **1** (2008) 1–23.

DE GROOT, P.A. (Ed.), Handbook of Stable Isotope Analytical Techniques, Vol. 1, Elsevier, Amsterdam (2004) 1258 pp.

EGERTON, R.F., Physical Principles of Electron Microscopy: An Introduction to TEM, SEM, and AEM, Springer, Boston (2005) 202 pp.

GILMORE, G.R., Practical Gamma-ray Spectrometry, 2nd edn, John Wiley and Sons, Chichester (2008) 387 pp.

HINDMAN, F.D., Actinide separations for alpha spectrometry using neodymium fluoride coprecipitation, Anal. Chem. **58** (1986) 1238–1241.

JONES, F.E., SCHOONOVER, R.M., Handbook of Mass Measurement, CRC Press, Boca Raton, FL (2002) 336 pp.

JOY, D.C., ROMIG JR, A.D., GOLDSTEIN, J.I. (Eds), Principles of Analytical Electron Microscopy, Springer, New York (1986) 448 pp.

KNOLL, G.F., Radiation Detection and Measurement, 4th edn, John Wiley and Sons, Chichester (2010) 864 pp.

LOURENCO, V., BOBIN, C., Weighing uncertainties in quantitative source preparation for radionuclide metrology, Metrologia **52** (2015) S18–S29.

LOWELL, S., SHIELDS, J.E., THOMAS, M.A., THOMMES, M., Characterization of Porous Solids and Powders: Surface Area, Pore Size and Density, Springer, Dordrecht (2004) 348 pp.

MAKISHIMA, A., Thermal Ionization Mass Spectrometry (TIMS): Silicate Digestion, Separation, and Measurement, Wiley-VCH, Weinheim (2016) 347 pp.

MERTZ, J., Introduction to Optical Microscopy, Roberts and Company Publishers, Greenwood Village, CO (2009) 413 pp.

MOORE, D.M., REYNOLDS JR, R.C., X-ray Diffraction and the Identification and Analysis of Clay Minerals, 2nd edn, Oxford University Press, Oxford (1997) 378 pp.

MOORE, E. (Ed.), Fourier Transform Infrared Spectroscopy (FTIR): Methods, Analysis and Research Insights, Nova Science Publishers, Hauppauge, NY (2016) 129 pp.

PONS, M.N., et al., Particle morphology: from visualisation to measurement, Powder Technol. **103** (1999) 44–57.

PROKIC, D., A method for high accuracy measurement of mass using modern electronic balances, J. Phys. E Sci. Instrum. **15** (1982) 282–284.

REED, S.J.B., Electron Microprobe Analysis and Scanning Electron Microscopy in Geology, 2nd ed, Cambridge University Press, Cambridge (2005) 232 pp.

SCOVIL, J.A., Photographing Minerals, Fossils, and Lapidary Materials, Geoscience Press, Tucson, AZ (1996) 224 pp.

SMITH, E., DENT, G., Modern Raman Spectroscopy: A Practical Approach, John Wiley and Sons, Chichester (2005) 210 pp.

TALVITIE, N.A., Electrodeposition of actinides for alpha spectrometric determination, Anal. Chem. **44** (1972) 280–283.

THOMAS, I.L., HAUKKA, M.T., XRF determination of trace and major elements using a single-fused disc, Chem. Geol. **21** (1978) 39–50.

TÖRÖK, P., KAO, F-J. (Eds), Optical Imaging and Microscopy, 2nd edn, Springer, Berlin (2007) 516 pp.

ANNEX I: SAMPLES AND GENERAL CONSIDERATIONS FOR NUCLEAR FORENSIC EXAMINATIONS

I-1. INTRODUCTION

A wide variety of nuclear and radioactive material and non-radioactive material may be the subject of a nuclear forensic examination. These materials may occur in various amounts, sizes and forms, with widely varying radioactive properties. Traditional forensic evidence contaminated with radionuclides may also be examined. Some considerations for the nuclear forensic examination of various types of nuclear or radioactive material sample are presented below.

I-2. SOLID SAMPLES

Solid samples that may be analysed as samples in a nuclear forensic examination may include nuclear material such as uranium ore, uranium fuel pellets and uranium or plutonium metals of varying isotopic composition. Radioactive samples, such as research reactor targets (irradiated or non-irradiated), radioactive scrap metal, sealed radioactive sources used for industrial or medical purposes and other associated materials, may also be analysed in a nuclear forensic examination. In principle, solid samples present a lower radioactive hazard to the analyst than powder and liquid samples because they do not present an inherent dispersible radioactive material hazard. However, laboratory personnel often consider the potential of inadvertently generating dispersible radioactive material from solid samples, especially during unpacking, handling and sub-sampling (aliquoting). Therefore, manipulation of solid samples may take place in a laboratory environment where dispersible radioactive material can be controlled, such as in a glove bag within a fume hood, or within a glove box. Aliquoting may present a particularly acute challenge, as cutting and grinding processes can easily produce radioactive particles. Physical measurements (such as measurement of sample dimensions) are a form of non-destructive analysis (NDA) technique and are typically taken before any other measurement, and the initial physical state of the solid sample can be documented through photography and optical microscopy.

For some NDA techniques, such as gamma spectrometry, sample preparation is often not performed for analyses of solid samples, provided the sample is small enough to fit in the instrument. In this case, many laboratories work to ensure that the sample is handled in such a manner as to not inadvertently produce dispersible radioactive material. Furthermore, measures can be taken to avoid contamination of the NDA instrument. Solid sample aliquoting is necessary for the use of most destructive analytical techniques, and the potential for sample heterogeneity is often considered when aliquoting a solid sample for analysis. Heterogeneity may occur at a variety of spatial scales and can be physical, chemical, elemental and/or isotopic in nature. Laboratory personnel can make use of data generated from NDA techniques to assess sample heterogeneity. Techniques such as optical microscopy and scanning electron microscopy (SEM) are especially useful for assessing the extent and degree of physical and elemental heterogeneity that may occur in a solid sample. It is more difficult to assess isotopic heterogeneity using these methods; therefore, it is often useful to prepare several aliquots for destructive analysis measurements of isotopic composition, such as mass spectrometry. Replicate destructive analysis measurements of different small aliquots can be used to assess isotopic homogeneity or heterogeneity in a solid sample; if the isotopic composition of several small aliquots of the sample, taken from different portions of the sample, overlap within measurement uncertainty, the analytical team may reach the conclusion that the sample is

isotopically homogeneous on the scale of the aliquots taken. Alternatively, spatially resolved analytical techniques capable of measuring isotopic composition, such as secondary ion mass spectrometry (SIMS) and laser ablation inductively coupled plasma mass spectrometry (ICP-MS), may be used to characterize sample isotopic heterogeneity on the microscale.

Sealed radioactive sources present an acute radiation hazard that, if mishandled, might cause severe injury or death to laboratory personnel. Most laboratories do not take the risk of breaching of the protective seal that contains the radioactive source, as the vast majority of useful nuclear forensic signatures can be measured using NDA techniques (such as gamma spectrometry), physical characterization using photography, measurement of physical dimensions and features, and perhaps microscopy. Breaching of the protective seal of a sealed radioactive source is typically performed only if absolutely necessary during the investigation, by trained personnel, within a facility capable of protecting laboratory personnel from gamma radiation and dispersible radioactive material that is likely to be generated.

Plutonium-bearing material also presents an acute radiation hazard, and many laboratories seek to detect the possible presence of plutonium, especially prior to aliquoting, which can generate dispersible radioactive material. As an alpha emitter, plutonium is extremely hazardous to laboratory personnel through inhalation and ingestion. Furthermore, contamination of the nuclear forensic laboratory with plutonium may render the laboratory unusable for a period of months or longer. Plutonium can be difficult to detect using gamma spectrometry instrumentation, especially if present with other, more highly active gamma emitters. Uranium-rich irradiated nuclear material may contain trace amounts of plutonium (as well as other activation products, in addition to fission products). Handling of plutonium and plutonium-bearing material is typically performed by trained laboratory personnel who are aware of the hazards and trained in the handling of dispersible radioactive material, using personal protective equipment to ensure containment of plutonium throughout the analyses.

I-3. POWDER SAMPLES

Powder materials that may be analysed as samples in a nuclear forensic examination include solid material encountered in powder form. Other radioactive material that may be encountered in powder form includes uranium ore concentrate, fine-grained material used in the nuclear fuel cycle (e.g. uranium tetrafluoride), contaminated ion exchange resins used in uranium purification processes and radionuclide contamination present on surfaces of non-nuclear material encountered at a radioactive crime scene. In many ways, powder samples are handled and analysed similarly to solid samples. However, powder samples present an inherent dispersible radioactive material hazard that many laboratories work to mitigate in order to protect laboratory personnel and the integrity of the laboratory itself. Many types of nuclear forensic sample may contain a particulate component, present as contamination on otherwise weakly radioactive contamination on solid materials does not contaminate the workspace or laboratory personnel.

Laboratory personnel may consider the possibility of heterogeneity at the individual grain scale, as powder samples may consist of grains or grain populations with widely varying physical, chemical, elemental and isotopic properties. For example, uranium-rich powders may contain grains with a range of isotopic compositions, from enriched to depleted. Uranium ore concentrate samples may contain grains of widely varying elemental and chemical composition and may include residual refractory grains suggestive of geological origin. Spatially resolved

imaging and measurement techniques, such as SEM, are invaluable for characterizing powder samples in the context of a nuclear forensic examination.

When aliquoting powders for use in destructive analysis, consideration is typically given to the amount of material taken for each aliquot. If the objective is to obtain a representative sample of the bulk composition of the powder, larger aliquots may be taken. Alternatively, laboratory personnel may seek to characterize the nuclear forensic signatures of different populations of grains present in the powder. In this latter case, it may be possible to separate out grains from representative populations, either manually or by exploiting some physical property of the particles themselves. Individual grains or grain populations can then be dissolved and analysed. Spatially resolved techniques, such as mass spectrometry with laser ablation sample introduction system, are particularly useful when examining a powder sample with complex populations of particles. With these techniques, it is possible to measure the elemental and isotopic composition of hundreds of individual grains, revealing heterogeneities that would otherwise be undetectable.

I-4. LIQUID AND SLURRY SAMPLES

On occasion, liquid radioactive samples may be encountered in the context of a nuclear forensic examination. In some ways, the handling of liquid samples is a simpler proposition than solids or powders, because liquids can be contained within sealed beakers, bottles or other similar vessels. However, the nature of the liquid matrix may vary, as various acids and solvents are used in the nuclear material production process. If transferring the liquid to a new vessel for storage and analysis (e.g. transfer to a beaker with standardized geometry for gamma spectrometry analysis), many laboratories aim to select a vessel composition that is compatible with the liquid matrix. For example, hydrofluoric acid-containing liquids are incompatible with glass; the use of polytetrafluoroethylene or perfluoroalkoxy alkane vessels may be preferred in this case.

A slurry is a mixture of undissolved solids suspended in and/or settled out of a liquid matrix. Radioactive slurries are common in nuclear facilities, particularly as waste products, and they present unique challenges in handling, aliquoting and analysis. There are several methods that may be used to separate the solid from the liquid components of a slurry, but it is usually necessary to consider the composition of the liquid component, as the separation method chosen might or might not be compatible with the liquid solvent or acid matrix. Centrifugation is a commonly used method to separate solid and liquid phases. Many laboratories take care to ensure that, once separated from the liquid matrix, the solid residue does not produce dispersible radioactive material.

I-5. SAMPLE DISSOLUTION METHODS

Solid and powder samples may involve dissolution (sometimes referred to as 'digestion', with the two terms being used interchangeably) prior to analysis, particularly using destructive analysis techniques. Dissolution of a solid sample involves the physical and chemical breakdown of the molecular structure and conversion to a solution (a homogeneous liquid mixture), typically performed using mineral acids (e.g. HCl, HF, HNO₃) or sometimes organic solvents. Upon dissolution, solid and powder samples can be analysed in solution form. Many destructive analysis techniques, such as mass spectrometry, can measure the elemental and isotopic properties of solutions with high precision. The dissolution of solid samples also enables more accurate and precise measurements using NDA techniques, such as gamma

spectrometry, because solution samples can be measured within containers of constant volume, eliminating the need for geometric correction. Accurate and precise masses for each solid sample aliquot are typically determined prior to dissolution in order to calculate elemental concentrations from measurements performed after digestion.

In nuclear forensics, inorganic materials such as metals, ores, oxides, ceramics and glass are frequently encountered. These materials can be dissolved using mineral acids, although the specific acid or acids most useful for this purpose are highly dependent on sample chemistry. For example, many uranium-rich materials are readily digestible using HCl and/or HNO₃. However, silicate-rich samples such as uranium ores are often best dissolved using HF to break silicon-oxygen bonds, ensuring that the sample is completely in solution. In some cases, H₂SO₄ can be used to ensure the sample aliquot remains in solution. In other cases, using a microwave digestion system can assist in ensuring complete dissolution of the sample. Many sample dissolution methods are described in the scientific literature, and a good first step when deciding on an approach to sample dissolution is to utilize a method that has been successful for the type of sample that is the subject of the nuclear forensic examination. Sample dissolution is sometimes a 'trial and error' process, and a sample is not always completely dissolved on the first attempt. Through subsequent cycles of chemical attack and evaporation, using different acids, solvents, and in some cases oxidizing or reducing agents, it is usually possible to completely dissolve the material.

Regardless of the dissolution method used, many laboratories work to ensure that the sample is not contaminated by dissolution reagents or the liquid vessels themselves. Laboratory personnel may acquire or prepare clean laboratory reagents, including mineral acids and water with minimal concentrations of trace metals. Beakers and other labware, such as disposable pipette tips and centrifuge tubs, may be cleaned of trace impurities using mineral acids and water prior to use, if not purchased at the trace-metal grade purity standard. Many laboratories take special care when samples are brought to evaporation between dissolution steps, because the potential for cross contamination might be high due to the ability of the residue to solidify and form particles that can exit their containers due to static electricity and/or air currents within the laboratory. Total procedural blank measurements can be performed alongside sample measurements to detect the presence of contamination that might impact data quality upon analysis.

I-6. CHEMICAL SEPARATION METHODS AND ISOTOPIC TRACERS

Some elemental and isotopic measurement techniques, such as alpha spectrometry and many forms of mass spectrometry, are optimally performed only when the element of interest has been chemically separated and purified prior to analysis. Chemical purification is typically performed only after complete digestion of the sample and can be performed using a wide variety of techniques, including solvent extraction, co-precipitation and ion exchange chromatography. Successful chemical separation and purification ensures that elements that have the potential to interfere with measurements are quantitatively removed. In mass spectrometry, the removal of interfering elements ensures that direct isobaric and polyatomic interferences do not cause erroneous measurements of the elements of interest. In alpha spectrometry, the production of well defined, narrow peaks is possible after chemical separation of the alpha emitting element of interest.

It is common to utilize isotopic tracers (often called 'spikes', with both terms being used interchangeably) in alpha spectrometry and mass spectrometry for accurate and precise

measurements of isotopic composition and elemental concentration, respectively, in a measurement approach known as isotope dilution. Isotopic tracers are commercially available solutions of high isotopic purity or of monoisotopic composition. In many applications, it is common to use spikes containing isotopes not expected to be present in the sample (e.g. ²³³U in a uranium-rich material). Mass spectrometry and alpha spectrometry techniques can determine isotope ratios with very high precision, and the use of tracers takes advantage of this benefit. It is possible to calculate the concentration of one or more isotopes of an element with high accuracy and precision by measuring the ratio of the spike isotope to the isotopes of interest. Many laboratories choose to measure an unspiked aliquot of the sample solution to perform the calculation as well. Many isotopic tracers used in nuclear forensics, such as for uranium and plutonium, are commercially available from producers of standard reference materials. In other cases, isotopic tracers can be manufactured and calibrated at the laboratory. For example, the short lived ²³³Pa isotopic tracer, which enables the precise and accurate measurement of ²³¹Pa for ²³⁵U–²³¹Pa age dating, can be continually produced and calibrated in the laboratory due to its half-life of approximately one month.

Although nuclear forensic analyses typically do not demand ultra-high precision for concentration measurements, radiochronometry measurements are greatly improved using this approach. Using mass spectrometry and well calibrated isotopic tracers, it is possible in some cases to obtain material ('production') age data with uncertainty as small as ± 1 year or less.

I-7. STANDARD REFERENCE MATERIALS

Standard reference materials are typically analysed alongside nuclear forensic samples to ensure that nuclear forensic data are accurate. Through the analysis of the properties of standard reference materials for which they have been certified (e.g. iron concentration, uranium isotopic composition), the laboratory analytical team can have higher confidence that the data generated for nuclear forensic samples are accurate. Many standard reference materials (such as solutions certified for concentration of specific elements) can be purchased from laboratory suppliers, and the nuclear industry produces standard reference materials relevant to nuclear measurements, such as those certified for ²³⁵U/²³⁸U and ²⁴⁰Pu/²³⁹Pu. Whenever possible, standard reference materials are analysed in the same batch of measurements as nuclear forensic sample preparation, standard reference materials are usually handled, digested and measured using the same procedures as nuclear forensic samples. For example, in the analysis of a uranium fuel pellet in a nuclear forensic examination, the laboratory team may acquire and analyse a uranium fuel pellet standard reference material which has been certified for uranium isotopic composition and radiochronometry age.

ANNEX II: DESCRIPTIONS OF ANALYTICAL METHODS USED IN NUCLEAR FORENSIC EXAMINATIONS

II-1. VISUAL INSPECTION AND PHOTOGRAPHY

II-1.1. Introduction and instrumentation considerations

Photographic documentation of samples helps ensure that their physical properties are recorded upon receipt by the laboratory prior to aliquoting and analysis. Photographic documentation can also be used throughout the handling, aliquoting and analysis process. Photography for nuclear forensic examination is typically performed using digital cameras, which can range in sophistication from mobile phone cameras to single lens reflex cameras with zoom lenses. The choice of camera is usually dependent primarily on the desired image resolution. For nuclear forensics, virtually any digital camera will suffice, provided photographs are taken in a systematic and well documented fashion. It is helpful to use a tripod or other stabilizer to ensure that images are not blurry or lacking focus. Lighting may be arranged so that the sample is well lit, and certified colour charts (e.g. the Macbeth ColorChecker) can be used to ensure that the colours captured by the photographs are accurate and can be traced to a standard reference.

II-1.2. Use in nuclear forensic examinations

In nuclear forensics, photography is typically performed upon initial receipt of the sample by the laboratory, especially during the unpacking and initial examination process. Use of a scale (e.g. ruler) enables quantitative measurement of specific features, dimensions and other physical properties of the sample. Photographs can record physical nuclear forensic signatures such as colour, texture, physical dimensions and evidence of handling (e.g. scratches, chips and other imperfections), as well as the homogeneous or heterogeneous physical nature of samples. Photography can be used throughout the nuclear forensic examination to record key details of the sample, as well as the methods used to perform analyses. During aliquoting, each subsample may be photographed prior to dissolution or alteration for destructive analysis.

II-2. OPTICAL LIGHT MICROSCOPY

II-2.1. Introduction and instrumentation considerations

Optical light microscopy is a non-destructive analysis (NDA) technique used to examine and record macro- and microscale physical features of nuclear forensic samples. Optical light microscopy, especially when paired with a digital imager, can be used to collect physical images of the surfaces of nuclear forensic samples. Many types of optical microscope are commercially available and may be purchased with varying magnification capabilities. Optical light microscopes are relatively simple to use and typically do not necessitate advanced training for their operation. Most optical microscopes with digital imagers can be purchased with image capture software, allowing for control over image properties such as contrast and white balance.

II-2.2. Use in nuclear forensic examinations

Optical light microscopy is particularly useful for identifying regions of interest in a nuclear forensic sample that can be characterized physically and later analysed in further detail using spatially resolved techniques, such as scanning electron microscopy (SEM), or subsampled during aliquoting. Physical heterogeneities and distinguishing features can often be readily identified using optical light microscopy. Optical light microscopy can be used during the process of aliquoting, particularly when removing small amounts of sample material for further analysis. Optical light microscopy can also be used to identify traditional forensic evidence that may be associated with nuclear or other radioactive material.

II-3. LABORATORY ANALYTICAL BALANCE

II-3.1. Introduction and instrumentation considerations

An analytical balance (sometimes referred to as a 'scale') is an instrument used to measure the mass of an object. Most laboratory analytical balances can measure the masses of objects in the gram to milligram range, although balances capable of measuring larger and smaller masses are also commercially available. Many analytical balances make use of a sample chamber that can be enclosed when a mass measurement is taken to ensure that air circulation in the room does not affect the quality of the measurement. Analytical chemistry laboratories are often equipped with one or more analytical balances, which are used in virtually all laboratory applications.

II-3.2. Use in nuclear forensic examinations

The mass of a sample is among the most fundamental measurements in a nuclear forensic examination and is typically the very first quantitative measurement performed (other than dose rate measurements for assessing the radiological hazard and accompanying photography). Analytical balances may be calibrated frequently to ensure measurement accuracy. Balance calibration can be performed using commercially available calibration weights. To establish measurement precision, many laboratories take several mass measurements of a sample if possible. Due to accumulation of static electricity, mass measurements can fluctuate, resulting in small but variable inaccuracies in mass measurement, especially for massing of plastics. Anti-static devices, either installed as a function of the balance itself or the use of an external device, can decrease fluctuations in mass measurement, resulting in increased mass accuracy and precision.

II-4. PROFILOMETRY AND THREE-DIMENSIONAL SCANNING

II-4.1. Introduction and instrumentation considerations

A variety of profilometry and three-dimensional (3D) scanning techniques can be used to analyse the surface and form of objects. Profilometry consists of a set of methods using optical light to qualitatively or quantitatively characterize the topography and relief of a sample surface. Photogrammetry and 3D scanning are methods for qualitative or quantitative characterization of volumes. These techniques are generally non-destructive, although some preparation techniques (e.g. application of reference marks) may alter the sample. Digital cameras can be used with photogrammetry software to manually generate 3D meshes with surface images, but the process is labour intensive if it is not automated. Modern profilometers use laser interferometry to generate two-dimensional (2D) or 3D reconstructions of material surfaces without touching the surface of the sample. Optical scanning usually involves little or no sample preparation. Samples with a matte surface and heterogeneous appearance with contrasting features are ideally suited to optical scanning. Reflective surfaces pose a challenge; lasers deflect from shiny surfaces, whereas photogrammetric techniques are impeded by reflections, which are difficult to distinguish from actual surface features.

II-4.2. Use in nuclear forensic examinations

In nuclear forensics, profilometry is used to characterize the surface of samples, which can provide insight into how materials were manufactured or otherwise produced. In cases where chipped or broken surfaces are identified, profilometry may aid in the determination of the tool or method used to alter the material. 3D scans may be used to determine sample volume, which may then be used to calculate material density and may reveal microscale physical features and structures not readily apparent in photographs.

II-5. PYCNOMETRY

II-5.1. Introduction and instrumentation considerations

Pycnometry is an NDA technique that typically makes use of a gas (or, less commonly, a liquid) to determine the displacement caused by a sample within a fixed volume, allowing for measurement of sample volume. This technique is ideal for determining the volume of irregularly shaped objects. If the sample mass is known (through measurement using an analytical balance), it is possible to calculate the sample density using volume data obtained through pycnometry.

II-5.2. Use in nuclear forensic examinations

Gas pycnometers can be used in nuclear forensics for determining the volume and density of a wide variety of solid materials, including uranium ore, uranium ore concentrate, fuel pellets, radioactive powders, glasses and many other materials. Many metallic materials common in the nuclear fuel cycle have characteristic densities (e.g. uranium metal has a characteristic density of 19.1 g/cm³), and the calculation of density can help determine the metallurgy of pure metals and alloys. However, samples with closed pore spaces within the volume of the sample can produce erroneous results. Many laboratories use a pycnometry gas that does not react with the sample, typically helium or another inert gas. Many laboratories take particular care while measuring the volume of powder samples by gas pycnometry, because it is possible to inadvertently disperse a radioactive powder when the sample chamber is flooded with gas.

II-6. SCANNING ELECTRON MICROSCOPY

II-6.1. Introduction and instrumentation considerations

SEM is a versatile form of microscopy that uses an electron beam to generate secondary electrons, backscattered electrons and X rays, which are detected and converted to images that can be used to characterize sample topography and elemental composition. SEM provides morphological images using secondary electron imaging as well as imaging of relative electron density (associated with the atomic number) from backscatter electrons. SEM is well suited to the observation of features on a nanometre to micrometre scale as it offers higher spatial resolution and depth of field in comparison to optical microscopy. Many SEM instruments are equipped with energy dispersive spectrometers for semi-quantitative characterization of elemental composition. Wavelength dispersive spectrometers are also increasingly common on modern SEM instruments, as well as electron microprobes. These spectrometers facilitate higher precision measurements of elemental concentrations and, in many conditions, can generate quantitative results. Most modern SEM instruments use secondary electron detectors capable of image resolution of a few nanometres and backscattered electron detectors for spatial distributions of material composition. SEM instruments optimized for elemental characterization are equipped with energy dispersive spectroscopy for rapid detection of X rays from elements excited by the electron beam or wavelength dispersive spectroscopy for detection and counting of individual characteristic X rays from a specific element. Image processing software allows for semi-quantitative assessment of the modal abundance of multiphase materials and automated calculations of grain size distribution, void space distribution and defect abundance. Several SEM software packages allow for quantitative determination of some morphological characteristics.

II-6.2. Use in nuclear forensic examinations

During a nuclear forensic examination, SEM can be used to characterize the morphology of materials. SEM can reveal grain size, shape and microstructures from the micro- to the nanoscale. For example, analysis of UO₂ fuel pellets can reveal the grain size distribution, grain shape, void size and shape, defects, homogeneity of the material, oxidation, cracking, traces of machining, porosity and other relevant properties. Morphological characteristics can be interpreted in the context of the processing history of the material, such as during the nuclear fuel fabrication process. The images of grain sizes of nuclear material can be used to perform comparative analyses between samples. In the case of uranium fuel pellets, the grain size may be related to the initial morphology, time and temperature that was applied to the material during sintering. The energy or wavelength dispersive spectroscopy capability of many SEM instruments provides elemental composition data, which can be quantitative when performed alongside measurements of standard reference materials. Compositional mapping can be a useful tool for measuring spatial distributions of major and trace elements in a bulk material. During a nuclear forensic examination, this technique is commonly applied to inorganic materials such as metals, oxides, minerals, glasses, semiconductors and ceramics.

II-6.3. Additional considerations

SEM imagery is often used for qualitative comparisons between nuclear forensic samples and materials of known production history. Current research in nuclear forensics is focused on developing quantitative methods of physical characterization from data obtained using SEM. Furthermore, research is being performed towards better understanding how particular processes in the nuclear fuel cycle impart specific physical nuclear forensic signatures to nuclear or other radioactive material, and how these nuclear forensic signatures can be interpreted in the context of nuclear forensic examinations.

Preparation for SEM can be challenging for some types of sample. Samples either need to be inherently conducting or, if they are not, covered with a thin conductive coating (e.g. carbon or gold). Variable pressure or environmental SEM instruments are available for analysis of some types of sample and can be used to avoid the need for conductive coatings. Quantitative compositional analysis involves instrumental calibration using standard reference materials. Many laboratories exercise caution when analysing nuclear forensic samples that have the potential to cause the release of dispersible radioactive material.

Capturing a representative set of images for a complex matrix can be difficult, depending on the degree of sample heterogeneity. Laboratory personnel with significant SEM expertise can provide guidance, assisted by subject matter experts in specific types of nuclear or other radioactive material. Quantitative image analysis is a complicated task and benefits from careful integration of sample preparation, imaging and image processing methods, particularly for powdered or particulate samples.

II-7. RAMAN SPECTROSCOPY

II-7.1. Introduction and instrumentation considerations

Raman spectroscopy is a spatially resolved NDA technique that can be used to determine the chemical composition of a sample. This technique induces low frequency vibrational transitions in molecules, resulting in the scattering of incident monochromatic light that reveals characteristic spectral patterns indicative of the crystalline phase or phases present in a sample. A typical Raman spectrometer consists of a monochromatic light source, a notch filter and a detection system, along with system-dependent optics including lenses and mirrors. Most systems make use of a range of laser wavelengths, including visible and infrared wavelengths to assist in the elimination of interferences caused by sample fluorescence. Use of the notch filter is useful for minimizing the Rayleigh scattering peak, which can overwhelm the desired Raman scattering peaks. Interpretation of spectra generated using this technique is generally performed through comparison against a database of the spectra of known molecular forms.

II-7.2. Use in nuclear forensic examinations

Raman spectroscopy is often used in nuclear forensics for identifying and determining the abundance of crystalline solids. In particular, this technique is useful for the determination of crystalline phases for uranium and plutonium oxide materials, such as UO_2 , UO_3 , U_3O_8 , PuO_2 and others. This technique is generally not suitable for the analysis of metallic samples and metal alloys. Most Raman spectrometers benefit from the use of spectral databases for data interpretation. Although these databases contain a wide variety of spectra of known crystalline materials, because nuclear forensic samples may be of unusual composition, it is possible that the crystalline forms of some samples might not be present in commercially available databases.

II-8. X RAY DIFFRACTION SPECTROSCOPY

II-8.1. Introduction and instrumentation considerations

X ray diffraction spectroscopy (XRD) is an NDA technique used to determine the mineralogical or molecular composition of crystalline and semi-crystalline materials. XRD instruments make use of a monochromatic X ray beam and X ray detector to produce a diffraction pattern over a range of diffraction angles. The diffraction pattern produced for each sample is a function of the characteristic d-spacings of the crystal lattices of the crystalline phases present. XRD can provide information on the composition, concentration, and purity of single- or multi-component crystalline materials. Typically, samples are analysed as fine-grained, loose powders mounted to produce a flat surface, although other preparations are possible, depending on the specific application of the technique.

II-8.2. Use in nuclear forensic examinations

XRD is widely used in nuclear forensics to determine the molecular composition of nuclear or other radioactive material, in particular uranium- and plutonium-rich materials such as UO_2 , U_3O_8 , metallic U, PuO_2 and other materials. XRD is particularly useful for determination of a range of crystalline compositions that may occur in uranium ore and uranium ore concentrate materials. Most XRD instruments include access to spectral databases when an instrument is procured; these databases contain various spectra of known crystalline materials. However, because nuclear forensic samples may be of unusual composition, it is possible that the crystalline forms of some samples might not be present in commercially available databases.

II-9. X RAY FLUORESCENCE SPECTROSCOPY

II-9.1. Introduction and instrumentation considerations

X ray fluorescence spectroscopy (XRF) is an analytical technique used to measure the elemental composition of solid and liquid materials. XRF instruments make use of a primary, monochromatic X ray beam to induce fluorescence in samples. The phenomenon of fluorescence results in the production of secondary X rays, which are characteristic of the elemental composition of the sample. Secondary X rays are measured by one or several detectors, depending on the instrument type. In energy dispersive spectrometry, a single detector is used to both discriminate and quantify the incident secondary X rays. In wavelengthdispersive spectrometry, a series of crystals are used to diffract the X rays of specific wavelengths, based on the known wavelengths of X rays produced by specific elements and the known d-spacings of analyser crystals built into the instrument. In either case, incident X ray intensities can be converted to elemental concentrations through semi-quantitative or quantitative conversion, using the fundamental parameters of X ray generation or through calibration with standard reference materials, respectively. XRF instruments can measure the concentration of most elements with analytical times of tens of minutes. Analyses can be performed non-destructively, provided the sample can fit into the sample chamber of the instrument, with a flat side available for analysis. While in principle XRF is an NDA technique, some applications of this technique benefit from sample preparation. Samples can be altered to optimize analysis as pressed pellets, glass beads or glass discs. Powder samples can also be measured by XRF if contained within a sample cup and supported by a membrane permeable to X rays.

II-9.2. Use in nuclear forensic examinations

XRF is a valuable technique for performing elemental composition analyses non-destructively and is often among the first elemental measurement techniques used in a typical nuclear forensic examination. XRF is particularly useful for a wide range of nuclear material and nonnuclear materials, such as metals, oxides, geological materials and other types of sample. Due to the non-destructive nature of XRF, samples can be initially measured using a semiquantitative analytical programme, identifying the major and minor elements present in the sample. If needed, additional quantitative measurements can be performed using XRF after additional sample preparation or using other analytical techniques, such as mass spectrometry.

II-9.3. Additional considerations

XRF instruments are less expensive, more user-friendly and more versatile than other elemental concentration analytical methods such as inductively coupled plasma mass spectrometry (ICP-MS). Modern X ray tubes have a lifespan of five or more years if maintained properly. Most XRF instruments can be operated by a technician with basic laboratory experience and training, although analyses of non-standard matrices may necessitate a higher level of expertise. XRF detection limits are in the order of 1–100 parts per million for most elements, limiting the utility of this technique for the analysis of trace constituents present in abundances below the detection limit. Elements lighter than sodium are difficult to analyse with most XRF instruments. Calibration standards do not exist for many sample matrices of interest. Interferences can be difficult or impossible to resolve in some cases (e.g. uranium spectral interference on molybdenum and zirconium). XRF data are usually less precise and sometimes less accurate than ICP-MS results. However, XRF is useful for measuring analytes that are difficult to analyse by ICP-MS, including chlorine, bromine, sulphur and phosphorous. Many samples can be analysed by XRF with little to no sample preparation. For more precise and accurate data, the analysis of a flat and polished face (solid samples) or fine powder is optimal. In cases where sample mass is limited, sample thickness may be a concern. In most cases, quantification of sample composition is performed under the assumption that X rays are generated from a sample that is 'infinitely' thick. In cases where only a thin layer of sample can be used $(0.1-0.01 \ \mu m)$, quantitative calculations will systematically underestimate heavy element concentrations. For analysis of thin samples, calibration curves can be determined through measurement of standard reference materials of similar matrices.

II-10. FOURIER TRANSFORM INFRARED SPECTROSCOPY

II-10.1. Introduction and instrumentation considerations

Fourier transform infrared spectroscopy (FTIR) is a spatially resolved technique for the determination of chemical composition. The technique involves the use of infrared light absorbance or emission to produce spectra which result from transitions in vibrational states in molecules present in the sample. Vibrational states are unique to specific functional groups or bonds within molecules, and can serve as a means of identifying unknown compounds and distinguishing between compounds having similar compositions. FTIR allows for rapid analysis of solids, liquids and gases at high spectral resolution (2 cm⁻¹).

In most FTIR instruments, a broadband infrared light source interacts with a sample (e.g. transmission or reflection), and light from the sample is directed to a Michelson interferometer. The Michelson interferometer consists primarily of a beam splitter, two mirrors and a photodetector. Incident light is partially reflected by the beam splitter towards one mirror and partially transmitted through the beam splitter to another mirror. The mirrors reflect the light back at the beam splitter, which recombines the light and partially directs it towards the photodetector. While one of the mirrors is fixed, the other mirror is designed to move over a set distance in such a way that, when the light recombines, it constructively and destructively interferes, resulting in varying intensity at the photodetector as a function of mirror position. This intensity pattern is processed by a fast Fourier transform, which converts it into an intensity spectrum as a function of wavelength. The vibrational modes of a molecule may interact with the incident broadband infrared light source if it experiences a change in dipole moment as it vibrates. Because of this, molecular oxygen and molecular nitrogen do not interact with the incident infrared radiation. Infrared frequencies that match the resonant vibrational frequency of molecular bonds may be absorbed and are thus absent (or suppressed) from the light that is directed into the Michelson interferometer. In infrared absorption spectroscopy, the absorbed frequencies appear at lower intensity in the wavelength spectrum. This resultant spectrum can be used to identify specific bonds and functional groups based on these observed vibrational modes.

II-10.2. Use in nuclear forensic examinations

FTIR is complementary to Raman spectroscopy in the types of vibrational mode that are observable by each technique. Specifically, for a molecule to be 'infrared-active', there needs to be a change in dipole moment, rather than a change in polarizability as necessitated by Raman spectroscopy. A multitude of infrared spectral libraries exist for common compounds, although a database specifically for nuclear forensics has not been developed. FTIR has been used as a means for identification and classification of uranium ore concentrates. Desktop and portable FTIR instruments are commercially available, depending on the application.

II-11. GROSS RADIATION COUNTING TECHNIQUES

II-11.1. Introduction and instrumentation considerations

Initial radiation surveys and radionuclide characterization in the laboratory can be performed using dose meters and contamination meters to determine the magnitude and location of the radiation field and/or the extent of the radioactive contamination. Ionizing radiation can be detected using Geiger-Müller instruments, ion chambers and gas proportional counters which, by interacting with electromagnetic radiation or charged particles, produce electrical signals that are converted to dose rate. Alternate methods for detection of alpha, beta and gamma radiation can also be performed through a variety of solid media, such as with silicon- and germanium-based detectors, cadmium telluride detectors or through scintillation in a solid-state detector such as sodium iodide and cadmium telluride. Liquid scintillation techniques can also be used for the measurement of some specific types of radionuclide, such as ³He or ¹⁴C. Neutron detection is typically performed using neutron absorbing materials, as neutrons do not directly produce ionizing radiation when interacting with matter. Neutron detectors typically make use of detector materials with high probabilities for low energy neutron absorption. Common commercially available portable units include gas proportional detectors filled with ³He and boron-based media in solid or gas form.

II-11.2. Use in nuclear forensic examinations

Measurements of dose rate and contamination levels are often performed in the field. Data from in-field measurements can be useful in the development of a nuclear forensic analytical plan. Once materials are received at the laboratory, measurements are often repeated for verification of the initial readings and integrity of package during transport. Radiation detectors are a critical part of safety procedures in a nuclear forensic laboratory, as they are used for monitoring the workspace and samples for contamination. These types of instrument provide an important service for understanding the presence of and potential risk posed by radiation. In addition, some types of handheld radiation equipment can perform preliminary identification of key radionuclides.

II-11.3. Additional considerations

Radiation detection equipment is typically calibrated using radioactive sources of known radiological properties to confirm measurement accuracy under normal operating conditions. In most cases, radiation detection measurements start with a measurement of background radiation levels. Most radiation detection instruments are sensitive to the ambient radiation background, local temperature, moisture and other environmental properties. When working in contaminated areas or locations with other radiation sources or background, laboratory personnel typically aim to shield their detectors and avoid instrument contamination. Some detectors are optimized for low level and/or low energy radiation and can become saturated in areas of high radiation. Nuclear forensic samples containing multiple radionuclides can create signals too complex to be correctly identified by portable detectors. Preliminary identification in the field is useful as a guide but can be confirmed through more robust means in a controlled laboratory environment.

II-12. GAMMA SPECTROMETRY

II-12.1. Introduction and instrumentation considerations

Gamma spectrometry is an NDA technique used to identify and quantify radionuclides which emit gamma radiation. This technique can be applied to a wide range of nuclear or other radioactive material or to evidence contaminated with gamma emitting radionuclides. Nuclear forensic samples can usually be analysed as received, without sample preparation. For more quantitative measurements, comparison to a standard reference material similar to the investigated item is necessary. For nuclear forensic characterization measurements, high resolution gamma spectrometry systems are often used. This type of system consists of a high purity germanium detector, cooled either by liquid nitrogen or by an electrical cooling system with electronic units, and a computer for displaying, saving and analysing gamma spectra. Low-background shielding made from lead, or another dense metallic material, is often used in order to isolate the detector from natural background radiation and interferences from other radioactive material. Coaxial high purity germanium detectors of a relative efficiency around 30% with beryllium or carbon windows are considered to be broadly applicable to most nuclear forensic applications. In order to maximize the amount of information obtained by gamma spectrometry, nuclear data libraries and software packages that can aid in activity and isotopic ratio determination might be needed. Additionally, a coaxial high purity germanium system can be complemented with a planar high purity germanium system, which can enable the measurements of radionuclides emitting low energy gamma rays (e.g. ¹²⁵I) with higher accuracy and precision. When measurements from coaxial and planar detectors are used together, the resolution and efficiency advantages may provide additional information about a sample.

II-12.2. Use in nuclear forensic examinations

Gamma spectrometry can be applied during both the categorization and characterization phases of the examination. During the categorization phase of nuclear forensic examination, gamma spectrometry can be used to achieve an initial identification of the radionuclides present in the sample. Gamma spectrometry measurements can then be performed in the laboratory using higher resolution instruments for longer count times. Further characterization measurements using gamma spectrometry can provide data on: (i) the isotopic composition of uranium or plutonium; (ii) the time elapsed since last chemical separation (the model age) of uranium or plutonium, which can help to narrow the possible origins of potential manufacturing facilities; (iii) nuclear forensic signatures of uranium reprocessing, which might link the sample to facilities handling reprocessed uranium; (iv) presence of cross contamination with other radionuclides, which can provide additional information on the production facility; and (v) quantitative determination of the total amount of uranium or plutonium.

II-12.3. Additional considerations

Gamma spectrometry measurements can be performed with little or no sample preparation prior to analysis. Gamma spectrometry is more cost effective than many destructive analysis techniques, and data can be generated in a matter of minutes to hours, although samples with low levels of activity can necessitate longer timescales of analysis. Measurement accuracy and precision can often be improved by using standard reference materials for efficiency and energy calibrations. However, the precision of gamma spectrometry is often lower than that of destructive analysis techniques. Relative uncertainties of at least a few percent are typical for absolute activity measurements and may be as low as 0.1–1% for isotope ratio measurements. Detection limits may vary by many orders of magnitude, depending on radionuclide, and may also depend on sample composition. Some radionuclides decay with very low gamma emission probabilities, resulting in large uncertainties. Some radionuclides do not emit gamma radiation, and therefore cannot be analysed using this technique. ²³⁶U, ²⁴²Pu or ⁹⁰Sr and ⁹⁰Y are examples of isotopes that may be encountered in nuclear forensic examinations of nuclear or other radioactive material that cannot be quantified by gamma spectrometry.

II-13. ALPHA SPECTROMETRY

II-13.1. Introduction and instrumentation considerations

Alpha spectrometry is a destructive analysis technique that can be used to measure the activity of radionuclides that decay via alpha emission. Alpha spectrometry involves dissolution and chemical separation prior to analysis. Purified alpha emitting radionuclides in solution can be deposited as thin films on filters prior to analysis. Alternatively, purified solutions can be electroplated prior to analysis. Samples are analysed by counting the number of alpha particles detected in a sample chamber, resulting in the production of alpha spectra that display peaks corresponding to the characteristic energies of the measured alpha particles. Measured alpha particles. Measured alpha peaks can be integrated, and their activity ratios can be calculated, which can then be converted to atomic ratios.

The type of detector used to quantify alpha emitting nuclides is usually a surface-barrier silicon detector in which alpha particles are deposited in the active volume of the detector. As alpha particles readily interact with matter, including air, the sample chamber is typically held under vacuum conditions. A slight positive pressure (~100 mtorr) is usually maintained to reduce detector contamination from recoiling daughter nuclides. Other detector types include Frisch grid detection systems (a type of gas-proportional counter) and liquid scintillation counters optimized for alpha measurements. These systems may be used as screening tools but typically do not provide the low detection limit and energy resolution needed for nuclear forensics.

II-13.2. Use in nuclear forensic examinations

Alpha spectrometry is the preferred technique for quantifying the abundance of alpha emitting radionuclides with half-lives less than a few hundred years (e.g. ²¹⁰Po, ²³²U, ²³⁸Pu and some isotopes of Am and Cm). Alpha spectrometry can also be used to determine the activity of longer lived actinides (e.g. ²³⁵U) as a more cost effective and sustainable technique relative to mass spectrometry, and these data can be used to calculate isotope ratios. Parent and daughter nuclides (e.g. ²³⁴U-²³⁰Th) may be quantified to determine radiochronometry age.

II-13.3. Additional considerations

Chemical separation and purification techniques are necessary for alpha spectrometry sample preparation. The methods used to isolate the element of interest are similar to those used to prepare samples for thermal ionization mass spectrometry (TIMS) or ICP-MS. Traditional alpha spectrometry is not able to resolve alpha energy peaks with less than ~50 keV difference such as ²³⁹Pu and ²⁴⁰Pu. In some cases, a combination of alpha spectrometry and mass spectrometry can be used to obtain the isotopic composition of Pu and U samples. Increased measurement precision may be obtained using detectors free from contamination and using longer counting times.
II-14. MODIFIED DAVIES-GRAY TITRATION TECHNIQUE FOR URANIUM CONCENTRATION

II-14.1. Introduction and instrumentation considerations

The modified Davies-Gray technique is an electrochemical titration technique for determining the concentration of uranium in solution. Particularly useful for samples with high uranium content, this technique is often used to determine the uranium concentration of dissolved solid uranium metals, alloys, oxides and other materials. Although individual modified Davies-Gray analyses can be performed manually, this technique is greatly aided by using commercially available auto-titration systems, which can be configured for the specific reagents needed. As a destructive analysis technique, the modified Davies-Gray technique is best performed after the complete dissolution of a sample aliquot sufficient to deliver 25–75 mg uranium to the solution (unless the sample is already a liquid). The analyst typically prepares a set of chemical reagents at the requisite concentrations, including the titrant $K_2Cr_2O_7$. The iron and vanadiumbearing reagents are usually prepared on the day of analysis due to their sensitivity to redox reactions.

II-14.2. Use in nuclear forensic examinations

In nuclear forensics, the modified Davies-Gray technique is often used to determine the concentration of uranium in uranium-rich materials, particularly uranium ore and uranium ore concentrate materials. The modified Davies-Gray technique can also be used for uranium fuel pellets and other materials in the fuel cycle containing tens of mg of uranium or more. Relative to other elemental concentration measurement techniques, the modified Davies-Gray technique is a cost effective option, and relative uncertainty of uranium concentration data using this technique can be less than 1%. Ideally, samples are analysed in triplicate to determine measurement uncertainty and to account for the possibility of a failed titration, which can occur for a number of reasons (e.g. stuck stirring bar, overshooting of the potentiometric end point). As the titrant $K_2Cr_2O_7$, contains hexavalent chromium, it is toxic and should be handled using the proper personal protective equipment.

II-15. INDUCTIVELY COUPLED PLASMA MASS SPECTROMETRY

II-15.1. Introduction and instrumentation considerations

ICP-MS is a highly sensitive and versatile technique used to measure the elemental and isotopic composition of materials dissolved in solution. Samples are typically introduced to the instrument in solutions produced from the dissolution of solid samples. Key attributes of ICP-MS include: (i) the ability to analyse most metallic and semi-metallic elements; (ii) detection limits as good as 1–10 parts per trillion for some elements; (iii) the ability to measure tens of samples within a multi-hour analytical routine; (iv) a wide dynamic concentration range for different analytes when measuring multiple analytes in a single routine (approximately 9 orders of magnitude).

In ICP-MS, samples are ionized in an argon plasma, isotopes of elements are separated based on mass-to-charge ratio using a mass analyser (either a quadrupole or magnetic sector), and relative abundances of isotopes are quantified using Faraday detectors or other types of detector. Fully quantitative element concentrations can be obtained by determining detector response for each analyte in a sample relative to the response generated using reference solutions with certified elemental concentrations. Instrumental background and blank measurement intensities should also be accounted for. The final accuracy and precision of the technique can be calculated based on the analysis of standard reference materials analysed in the same routine as samples. The relative uncertainty of elemental concentration measurements by quadrupole ICP-MS can often be better than 5% relative uncertainty for each element. Isotope ratios can be measured with relative uncertainty better than 1% in many cases, depending on the type of ICP-MS used.

II-15.2. Use in nuclear forensic examinations

Quadrupole ICP-MS is useful for determining the concentration of virtually all metallic elements in a sample. ICP-MS instruments can also be used to measure isotopic composition of some key elements in nuclear forensic samples, such as uranium and plutonium. The triple-quadrupole ICP-MS can be used to provide useful isotope ratios with precision better than 0.1%, while sector field ICP-MS (equipped with single or multi collector) can provide precision lower than 0.01%. By measuring parent-daughter nuclide ratios, determination of sample age can be accomplished via ICP-MS measurements using radiochronometry.

II-15.3. Additional considerations

Quadrupole ICP-MS instrument manufacturers offer benchtop platforms and are less expensive than other types of mass spectrometer. Many models offer useful features such as collision or reaction cells, and some offer multiple quadrupole systems in order to reduce interferences during the measurement. There are several commercial vendors that produce sector field ICP-MS, with similar basic designs and performance. The detector configurations for these types of instrument are customizable at the time of purchase depending on the specific application the instrument will be used for. For example, the addition of multiple ion counters and abundance sensitivity filters can increase the precision of actinide isotopic analyses. Changes to the sample introduction system can result in significant performance enhancements. In particular, the use of high sensitivity sample and skimmer cones together with a desolvating nebulizer can result in an order of magnitude increased sensitivity. In many laboratories, for precise isotopic analysis, the use of desolvating nebulizers is now the most common operating configuration.

ICP-MS (quadrupole or sector field) offers a large flexibility in analysing a wide range of samples, with low detection limits (parts per billion or better) and high precision. The ease of use and versatility of the quadrupole ICP-MS is in part why it has gained popularity in the nuclear forensic analytical community. However, while quadrupole ICP-MS is a relatively inexpensive instrument applied to elemental and isotopic analysis, the more expensive sector field ICP-MS (equipped with single or multi-collector) or TIMS can generate higher precision isotope ratio data. Sector field instruments, especially multi-collector ICP-MS instruments, necessitate higher operational skills, higher maintenance, and higher operational and service costs.

Interferences from high concentrations of polyatomic compounds that may be present in the sample matrix, especially heavy elements, can reduce overall sensitivity and can result in inaccurate measurements. Interferences can be challenging to deal with, even for experienced operators. In addition, contamination of the sample introduction system can cause memory effects, increasing instrumental background. Advances in instrument design, including the introduction of collision and reaction cells in quadrupole ICP-MS, high resolution of sector instruments to manage interferences and aerosol dilution to handle heavy matrices, have enabled analysts to overcome some of these challenges.

Samples are primarily introduced to ICP-MS as solutions, although solid samples can also be analysed in-situ by coupling a laser ablation system to a mass spectrometer. Due to the sensitivity of the technique, sample consumption is small, and a typical sample analysis may consume less than a milligram of mass. To obtain higher quality isotopic data, sample solutions are usually purified before analysis to remove other elements (i.e. all elements other than the element of interest), including potential interfering species. This is usually achieved by ion exchange chromatography, using separation procedures tailored to the element of interest. Dissolved samples are usually introduced to the plasma in clean and dilute mineral acid solutions, such as 2% HNO₃.

II-16. INDUCTIVELY COUPLED PLASMA ATOMIC EMISSION SPECTROMETRY

II-16.1. Introduction and instrumentation considerations

Inductively coupled plasma atomic emission spectrometry (ICP-AES) is a destructive analysis technique used to determine elemental concentration using characteristic visible light electromagnetic radiation generated during sample ionization in a plasma source. As sample atoms cool following excitation from a plasma, thermally excited electrons emit photons with wavelengths characteristic of elemental composition, which are then detected. Although ICP-AES can measure the concentration of many elements, spectral interferences do occur and can impact the accuracy and precision of measurements of certain elements. However, ICP-AES instruments are robust in dealing with samples dissolved in atypical matrices and containing higher levels of dissolved solids. Therefore, ICP-AES is useful for analysing many kinds of dissolved materials.

II-16.2. Use in nuclear forensic examinations

In nuclear forensics, ICP-AES is a relatively cost effective method for measuring the elemental concentration of many elements without the need for chemical separation. As a destructive analysis technique, ICP-AES requires solid samples to be dissolved prior to analysis, necessitating accurate and precise measurement of aliquot mass prior to analysis. ICP-AES detection limits are in the parts per million range for many elements, and therefore may be unsuitable for some trace elements in nuclear forensic samples. As with any technique involving sample digestion, ICP-AES produces liquid radioactive waste that has to be handled safely and securely. Furthermore, the analysis of radioactive material has the potential to contaminate the instrument itself. Many laboratories make use of a sample introduction system used only for nuclear forensic samples when performing a nuclear forensic examination. Although some laboratories may prefer to use the more versatile ICP-MS technique, ICP-AES remains suitable for many nuclear forensic applications.

II-17. THERMAL IONIZATION MASS SPECTROMETRY

II-17.1. Introduction and instrumentation considerations

TIMS is a destructive analysis technique used to measure the isotopic composition of materials with high precision. For some applications, TIMS is considered the 'gold standard' for isotope ratio measurement, capable of producing ²³⁵U/²³⁸U measurements of better than 0.05%. Samples for TIMS are typically first dissolved, then purified using chemical separation techniques. After purification, samples are then carefully loaded onto filaments composed of rhenium, tantalum or tungsten. Filaments are loaded into an evacuated sample chamber and subjected to a current, which heats, evaporates and ionizes the sample. The ion beam is than accelerated using ion lenses into a magnetic sector, where the beam is separated based on mass-to-charge ratio. Most modern TIMS instruments are multi-collector instruments, enabling simultaneous detection of each isotope for high precision measurements.

II-17.2. Use in nuclear forensic examinations

While most nuclear forensic applications do not need the level of measurement precision offered by TIMS, this technique can be used to determine the isotopic composition of uranium and plutonium in nuclear forensic samples. In addition, TIMS is capable of precise measurements of other isotopic systems, including lead, neodymium, strontium and others. These measurements are most ideally suited to linking uranium ore or uranium ore concentrate materials to a geological origin, or to the precise determination of specific fission and activation products in irradiated nuclear fuel cycle materials.

II-18. SECONDARY ION MASS SPECTROMETRY

II-18.1. Introduction and instrumentation considerations

Secondary ion mass spectrometry (SIMS) is a spatially resolved analytical technique used for determining the isotopic composition of materials with high precision. Using SIMS, it is possible in some cases to determine the isotopic composition of some elements on a spatial scale of 1×10^{-6} m. Although nominally non-destructive, most samples need some degree of sample preparation prior to analysis, and the analysis itself results in the removal of nano- to micrometre-scale layers from the sample. Most SIMS instruments make use of a primary beam composed of either oxygen or caesium, which is focused on the flat surface of a sample. Atoms of the sample are then sputtered from the surface, ionized, accelerated through a sector field mass analyser and detected using an array of detectors. The SIMS technique is expensive and often makes use of technical personnel with extensive experience in the technique for operation.

II-18.2. Use in nuclear forensic examinations

Although widely used in nuclear safeguards applications, SIMS also has applications in nuclear forensics. SIMS has been used to determine spatial variations in uranium isotopic composition in uranium fuel pellets and other nuclear fuel cycle materials and is particularly useful for revealing microscale heterogeneity in samples. Furthermore, SIMS can be used to analyse the isotopic composition of uranium particles on the scale of 1×10^{-6} m, which are typically too small to analyse individually using bulk analytical techniques, such as inductively coupled plasma mass spectrometry.

II-19.STABLE ISOTOPE RATIO MASS SPECTROMETRY USING GAS SOURCE INSTRUMENTS

II-19.1. Introduction and instrumentation considerations

Stable isotope ratio mass spectrometry is a destructive analysis technique used to measure the isotopic composition of light stable isotopic systems, such as hydrogen, nitrogen, oxygen, carbon and sulphur. The stable isotope ratio mass spectrometer typically consists of a sample introduction system (with the sample often introduced as a gas after purification), a mass analyser and a detector or set of detectors. Many stable isotope ratio mass spectrometer systems use a dual inlet configuration. The dual inlet stable isotope ratio mass spectrometer employs two variable volume bellows. One bellows holds a reference gas, and the other bellows holds the sample gas. The bellows are adjusted so that the pressure is equal, and a switching valve is used to alternate between the reference and sample gases to obtain multiple measurements of the isotopic ratios. The dual inlet system allows for high precision measurements but involves off-line preparation of a pure gas from the original sample of interest. Typically, this preparation involves vacuum line chemistry and purification processes.

Continuous flow stable isotope ratio mass spectrometry has lower sample size demands and increases sample throughput using automated sample preparation peripherals. Continuous flow stable isotope ratio mass spectrometer uses a continuous gas stream, usually helium, to carry the sample gas into the stable isotope ratio mass spectrometer inlet. Compared with dual inlet stable isotope ratio mass spectrometer systems, continuous flow systems are typically less precise. However, continuous flow stable isotope ratio mass spectrometer provides a significant advantage by allowing for sample gas preparation to be performed online through interfacing with additional peripheral instruments. Gases in a mixture may be separated and purified by gas chromatography columns prior to introduction at the stable isotope ratio mass spectrometer inlet. Solids and liquids can be combusted or pyrolyzed to make a gas using an automated peripheral instrument, which significantly reduces sample preparation time. This has led to approaches combining stable isotope ratio mass spectrometer with peripherals such as elemental analysers, gas chromatography followed by combustion and liquid chromatography.

II-19.2. Use in nuclear forensic examinations

In nuclear forensics, stable isotope ratio mass spectrometer can be used to analyse the isotopic composition of light stable isotopes in samples. Nuclear material can have light element components, including crystalline materials like UO₂, U₃O₈, (NH₄)₂U₂O₇, U(SiO₄)_{1-x}(OH)_{4x} and UTi₂O₆, as well as glasses and other non-crystalline materials. Some laboratories are conducting ongoing research into the measurement and interpretation of light stable isotope nuclear forensic signatures, and interpretation of these nuclear forensic signatures may be complex.

ABBREVIATIONS

Fourier transform infrared spectroscopy
International Atomic Energy Agency
Inductively coupled plasma atomic emission spectrometry
Inductively coupled plasma mass spectrometry
International Organization for Standardization / International
Electrotechnical Commission
Non-destructive analysis
Nuclear Security Series
Scanning electron microscopy
Secondary ion mass spectrometry
Thermal ionization mass spectrometry
X ray diffraction
X ray fluorescence



ORDERING LOCALLY

IAEA priced publications may be purchased from the sources listed below or from major local booksellers.

Orders for unpriced publications should be made directly to the IAEA. The contact details are given at the end of this list.

NORTH AMERICA

Bernan / Rowman & Littlefield

15250 NBN Way, Blue Ridge Summit, PA 17214, USA Telephone: +1 800 462 6420 • Fax: +1 800 338 4550 Email: orders@rowman.com • Web site: www.rowman.com/bernan

REST OF WORLD

Please contact your preferred local supplier, or our lead distributor:

Eurospan Group

Gray's Inn House 127 Clerkenwell Road London EC1R 5DB United Kingdom

Trade orders and enquiries:

Telephone: +44 (0)176 760 4972 • Fax: +44 (0)176 760 1640 Email: eurospan@turpin-distribution.com

Individual orders: www.eurospanbookstore.com/iaea

For further information:

Telephone: +44 (0)207 240 0856 • Fax: +44 (0)207 379 0609 Email: info@eurospangroup.com • Web site: www.eurospangroup.com

Orders for both priced and unpriced publications may be addressed directly to:

Marketing and Sales Unit International Atomic Energy Agency Vienna International Centre, PO Box 100, 1400 Vienna, Austria Telephone: +43 1 2600 22529 or 22530 • Fax: +43 1 26007 22529 Email: sales.publications@iaea.org • Web site: www.iaea.org/publications

International Atomic Energy Agency Vienna